
%\documentstyle[12pt]{article}
% $Id: thesis.tex,v 1.1 1994/04/03 22:05:02 han Exp han $
%\textwidth=6in
%\textheight=9in
%\topmargin=-1in
%\oddsidemargin=0in
%\evensidemargin=0in

% draft format
%\documentstyle[11pt]{article}
%\voffset=-3cm
%\hoffset=-1cm
%\addtolength{\textheight}{5.5cm}
%\addtolength{\textwidth}{2cm}
%\renewcommand{\baselinestretch}{1.1}
%\sloppy

% graduate school format: double space, 1in margins, left +.5in
\documentstyle[11pt]{article}
\voffset=-2cm
\hoffset=-0.5cm
\addtolength{\textheight}{4.0cm}
\addtolength{\textwidth}{2cm}
\renewcommand{\baselinestretch}{2.0}
\sloppy

% landscape mode: set twocolumn style and use dvips: formulas to wide!
%\documentstyle[twocolumn]{article}
%\renewcommand{\baselinestretch}{1.0}
%\textwidth=10.5in
%\textheight=7.3in
%\columnsep=.5in
%\topmargin=-1in
%\oddsidemargin=-.75in
%\evensidemargin=-.75in
%\special{landscape}

\newcommand{\beq}{\begin{equation}}
\newcommand{\eeq}{\end{equation}}
\newcommand{\beqa}{\begin{eqnarray}}
\newcommand{\eeqa}{\end{eqnarray}}
\newcommand{\di}{\displaystyle}

\newcommand{\tr}{\mbox{tr}}
\newcommand{\pexp}{\mbox{Pexp}}
\newcommand{\fot}{\frac{1}{2}}
\newcommand{\fof}{\frac{1}{4}}
\def\a{\alpha}
\def\b{\beta}
\def\c{\gamma}
\def\d{\delta}
\def\e{\epsilon}
\def\subtilde#1{#1\llap{\lower2ex\hbox{$\widetilde{\hphantom{#1}}$}}}

\typein{Enter 'includeonly': }
%\includeonly{sec5,sec6}



\begin{document}
\begin{thebibliography}{99}
\addcontentsline{toc}{section}{References}
% my own reading
  \bibitem{Bruegman} 	
	Bernd Bruegman, 	
	Ph.D. Thesis, Syracuse University.  
  \bibitem{BrinkHenn} 	
	Lars Brink and Marc Henneaux,
	Principles of String Teory, 	
	Plenum Press 1988.  
  \bibitem{Goddard}
	P. Goddard, J. Goldstone, C. Rebbi, and C.B. Thorn, 	
	Nucl. Phys. {\bf B56} (1973) 109.  
  \bibitem{Brouwer} 	
	R.C. Brouwer, 	
	Phys. Rev. {\bf D6} (1972) 1655; 
	P. Goddard, and C.B. Thorn, 	
	Phys. Lett. {\bf 40B} (1972) 235.  
  \bibitem{Fujikawa} 	
	K. Fujikawa, 
	Phys. Rev. {\bf D25} (1982) 2584; 
	M. Kato and K.  Ogawa, 
	Nucl Phys. {\bf B212} (1983) 443; 
	S. Hwang, 
	Phys. Rev. {\bf D28} (1983) 2614.
  \bibitem{NeveuSchwarz}
	A. Neveu and J. H. Schwarz,
	Nucl. Phys. {\bf B31} (1971) 86;
	Phys. Rev. {\bf D4} (1971) 1109.
  \bibitem{Zhaba}
	Z. Haba, J. Sobczyk,
	Functional Integration, Geometry and Strings, 1988 (book).
  \bibitem{NambuGoto}
	see \cite{BrinkHenn}, page 289 ref 1,2
  \bibitem{GreenGross}
	M. Green and D. Gross,
	Unified String Theories (book)
  \bibitem{Wali}
	K. C. Wali
	Eighth Workshop on Grand Unification, 1987, Syracuse
  \bibitem{Kaku} 	
	M. Kaku,
	Introduction to Superstrings (book) 1988.
  \bibitem{KAWAI}
    H. Kawai and M. Ninomiya,
    Nucl. Phys. {\bf B336} (1990) 115.
    about random surfaces, extrinsic curv, Hausdorff, quantum gravity,
    beta function.
  \bibitem{POLEXT}
    A. M. Polyakov,
    Nucl. Phys. {\bf B268} (1986) 406.
  \bibitem{DYTRAS}
    V. A. Kazakov,
    Phys. Lett. {\bf 150B} (1985) 282;
    F. David,
    Nucl. Phys. {\bf B257} (1985) 45;
    J. Ambj{\o}rn, B. Durhuus and J. Fr\"{o}lich,
    Nucl. Phys. {\bf B257} (1985) 433.
	about dynamic triangulation ...
  \bibitem{IRBACK}
    A. Irb\"{a}ck, J. Jurkiewicz and S. Varsted,
    Computer Phys. Comm. {\bf 70} (1992) 59.
	(I have this)
  \bibitem{BALLIE}
    C. Baillie, D. Johnston and R. Williams,
    Computer Phys. Comm. {\bf ?} (year?) ?.
	(to find this)
  \bibitem{AMBTWO}
    J. Ambj{\o}rn, A. Irb\"{a}ck, J. Jurkiewicz and B. Petersson,
    Nucl. Phys. {\bf B393} (1993) 571.
    (I have this:  {\em The Theory of Dynamical Random Surfaces with
     Extrinsic Curvature}, pretty thick paper, many discussions about
     relation between string theory and random surfaces, numerical 
      measurements)
  \bibitem{DAVID}
    Nucl. Phys. {\bf B} (Proc. Suppl.) {\bf 17} (1990) 51
    F. David.
  \bibitem{KPZ}
    V. G. Knizhnik, A. M. Polyakov and A. B. Zamolodchikov,
    Mod. Phys. Lett. {\bf A3} (1988) 819.
	(famous one: exact solution for continuum model)
	(do I have this ?)
  \bibitem{KAZAKOV}
	V. A. Kazakov,
	Mod. Phys. Lett {\bf 119A} (1986) 140.
	(famous one: exact solution for discretized model)
	(do I have this ?)
  \bibitem{AMBONE}
    J. Ambj{\o}rn, J. Jurkiewicz, S. Varsted, A. Irb\"{a}ck
    and B. Petersson,
    Phys. Lett. {\bf 275B} (1992) 295.
		(I may have this in preprint, double check in lib,
very good many discussion about RS and theory aspects)
  \bibitem{Renken} 
	R. K. Renken, 	
	Nucl. Phys. {\bf B} (Proc. Suppl.) 20 (1991) 720.  	
	(many good discussion about properties of the crumpling
transition, I have this).
  \bibitem{Myers} 
	E. Myers, Bryce ...(need to add more names here),
    Nucl. Phys. {\bf B} (Proc. Suppl.) 20 (1991) 744.		
  \bibitem{BCHHM}
    M. Bowick, P. Coddington, L. Han, G. Harris and E. Marinari,
    Nucl. Phys. {\bf B394} (1993) 791 and references therein.
  \bibitem{JPOT}
	J. Potvin
	Computers in Phys. Vol. 7, No. 2 (Mar/1993) 149
  \bibitem{WHEATER}
	J. F. Wheater,
	Nucl. Phys. {\bf B} (Proc. Suppl.) 34 (1994) 15.

% from other paper ...
  \bibitem{DAVREV}
    See F. David,
    {\em Introduction to the Statistical Mechanics of Random Surfaces
      and Membranes}, Lectures given at the Eighth Jerusalem Winter
    School for Theoretical Physics, {\em Two Dimensional Gravity
    and Random Surfaces},
    Saclay preprint SPhT/91-113 and references therein.
  \bibitem{DADIKA}
    F. David,
    Mod. Phys. Lett. {\bf A3} (1988) 1651;
    J. Distler and H. Kawai,
    Nucl. Phys. {\bf B321} (1989) 509.
  \bibitem{KAZMIG}
    V. A. Kazakov and A. A. Migdal,
    Nucl. Phys. {\bf B311} (1988/89) 171.
  \bibitem{EXACT}
    E. Br\'{e}zin and V.A. Kazakov,
    Phys. Lett. {\bf 236B} (1990) 144;
    M.R. Douglas and S.H. Shenker,
    Nucl. Phys. {\bf B335} (1990) 635;
    D. J. Gross and A. A. Migdal,
    Phys. Rev. Lett. {\bf 64} (1990) 127.
    D. Gross and A. Migdal, 
    Nucl. Phys. {\bf B340} (1990) 333.
  \bibitem{KUTSEI}
    D. Kutasov and N. Seiberg,
    Nucl. Phys. {\bf B358} (1991) 600.
  \bibitem{BILDAV}
    A. Billoire and F. David,
    Phys. Lett. {\bf 168B} (1986) 87;
    Nucl. Phys. {\bf B275} (1986) 617.
  \bibitem{AMNOEX}
    J. Ambj{\o}rn, B. Durhuus and J. Fr\"{o}hlich,
    Nucl. Phys. {\bf B275} (1986) 161;
    J. Ambj{\o}rn, B. Durhuus, J. Fr\"{o}hlich and P. Orland,
    Nucl. Phys. {\bf B270} (1986) 457.
  \bibitem{EALRYM}
    D. Boulatov, V. Kazakov, I. Kostov and A.A. Migdal,
    Phys. Lett. {\bf 157B} (1985) 295.
  \bibitem{MINOEX}
    D. Boulatov, V. Kazakov, I. Kostov and A.A. Migdal,
    Phys. Lett. {\bf 174B} (1986) 87;
    Nucl. Phys. {\bf B275} (1986) 641.
  \bibitem{AMNEED}
    J. Ambj{\o}rn and B. Durhuus,
    Phys. Lett. {\bf 188B} (1987) 253.
  \bibitem{CATES}
    M.E. Cates,
    Europhys. Lett. {\bf 7} (1988) 719.
  \bibitem{POLBOO}
    A. M. Polyakov,
    {\em Gauge Fields and Strings}
    (Harwood Academic Publishers, Chur, Switzerland 1987).
  \bibitem{CANHAM}
    P. B. Canham,
    J. Theor. Biol. {\bf 26} (1970) 61.
  \bibitem{HELFRI}
    W. Helfrich,
    J. Naturforsch. {\bf 28C} (1973) 693.
  \bibitem{KLEINA}
    H. Kleinert,
    Phys. Lett. {\bf 174B} (1986) 335.
  \bibitem{WIEGMA}
    P. Wiegmann,
    Nucl. Phys. {\bf B323} (1989) 330.
  \bibitem{CHOBRU}
    Y. Choquet-Bruhat, C. DeWitt-Morette and M. Dillard-Bleick,
    {\em Analysis, Manifolds, and Physics}
    (Elsevier, Amsterdam 1987).
  \bibitem{DOCARM}
    M. P. Do Carmo,
    {\em Dif\-feren\-tial Geo\-metry of Cur\-ves and Sur\-faces}
    (Prentice-Hall, Englewood, N.J., USA 1976).
  \bibitem{POLSTR}
    A. M. Polyakov,
    Phys. Lett. {\bf 103B} (1981) 207.
  \bibitem{CATTER}
    S. Catterall,
    Phys. Lett. {\bf 220B} (1989) 207.
  \bibitem{BAJOWI}
    C. Baillie, D. Johnston and R. Williams,
    Nucl. Phys.  {\bf B335} (1990) 469.
  \bibitem{BCJW}
    C. Baillie, S. Catterall, D. Johnston and R. Williams,
    Nucl. Phys. {\bf B348} (1991) 543.
  \bibitem{CAKORE}
    S. Catterall, J. Kogut and R. Renken,
    Nucl. Phys. {\bf B} (Proc. Suppl.) {\bf 99A} (1991) 1.
  \bibitem{POLYAN}
    J. Polchinski and Z. Yang, {\em High Temperature Partition
    Function of the Rigid String},
    U. Texas and Rochester preprint
    UTTG-08-92/UR-1254/ER-40685-706 (hep-th/9205043).
  \bibitem{BRAZAC}
    E. Braaten and C. Zachos, 
    Phys. Rev. {\bf D35} (1987) 1512.
  \bibitem{KLEINB}
    H. Kleinert, Phys. Lett. {\bf 114A} (1986) 263.
  \bibitem{FORSTE}
    D. Forster,
    Phys. Lett. {\bf 114A} (1986) 115.
  \bibitem{PELLEI}
    L. Peliti and S. Leibler,
    Phys. Rev. Lett. {\bf 54} (1985) 1690.
  \bibitem{MORRIS}
    T. Morris,
    Nucl. Phys. {\bf B341} (1990) 443;
    J. Govaerts,
    Int. J. Mod. Phys.  {\bf A4} (1989) 173.
  \bibitem{POLSTRO}
    J. Polchinski and A. Strominger,
    Phys. Rev. Lett. {\bf 67} (1991) 1681.
  \bibitem{DIST3D}
    J. Distler, {\em A Note on the 3D Ising Model as a String Theory},
Prince\-ton Uni\-ver\-sity pre\-print
    PUPT-1324 (hep-th/9205100).
  \bibitem{DAGUAE}
    F. David,
    Europhys. Lett. {\bf 2} (1986) 577;
    F. David and E. Guitter,
    Europhys. Lett. {\bf 3} (1987) 1169;
    Nucl. Phys. {\bf B295} (1988) 332;
    F. Alonso and D. Espriu,
    Nucl. Phys. {\bf B283} (1987) 393.
  \bibitem{YANG}
    Z. Yang, {\em A Remark on Conformal Anomaly and Extrinsic
    Geometry of Random Surfaces}, Univ. of Rochester preprint
    UR-1136/ER-13065-678 (hep-th/9111003).
  \bibitem{CEKR}
    S. Catterall, D. Eisenstein, J. Kogut and R. Renken,
    Nucl. Phys. {\bf B366} (1991) 647.
  \bibitem{KROGOM}
    D. Kroll and G. Gompper,
    Science {\bf 255} (1992) 968;
    Europhys. Lett. {\bf 19} (1992) 581;
    {\em The Scaling Behavior of Randomly Triangulated Self-Avoiding
         Surfaces,} to appear in
    Phys. Rev. {\bf A}.
  \bibitem{NELSON}
    D. Nelson,
    in {\em Statistical Mechanics of Membranes and Surfaces},
    Jerusalem Winter School, Vol. {\bf 5},
    edited by D. Nelson, T. Piran and S. Weinberg
    (World Scientific, Singapore 1989) 167.
  \bibitem{DAVREZ}
    F. David,
    in {\em Statistical Mechanics of Membranes and Surfaces},
    Jerusalem Winter School, Vol. {\bf 5},
    edited by D. Nelson, T. Piran and S. Weinberg
    (World Scientific, Singapore 1989) 158.
  \bibitem{ENZOHI}
    M. Falcioni, E. Marinari, M. L. Paciello, G. Parisi and B. Taglienti,
    Phys. Lett. {\bf 102B} (1981) 270;
    Nucl. Phys. {\bf B190} (1981) 782;
    Phys. Lett. {\bf 108B} (1982) 331;
    E. Marinari,
    Nucl. Phys. {\bf B235} (1984) 123.
  \bibitem{SWEFEA}
    A. M. Ferrenberg and R. H. Swendsen,
    Phys. Rev. Lett. {\bf 61} (1988) 2635
    and {\em Erratum}, ibid. {\bf 63} (1989) 1658.
  \bibitem{ZINJUS}
	See, for example,
	J. Zinn-Justin, {\it Quantum Field Theory and Critical Phenomena}
	(Oxford University Press, New York, 1989).
  \bibitem{WOLFFO}
	U. Wolff, Nucl. Phys. {\bf B322} (1989) 759;
	Nucl. Phys. {\bf B334} (1990) 581.
  \bibitem{SWEFEB}
    A. M. Ferrenberg and R. H. Swendsen,
    Phys. Rev. Lett. {\bf 63} (1988) 1196.
  \bibitem{ALBESA}
    N. A. Alves, B. A. Berg and S. Sanielevici,
    {\em Spectral Density Study of the $SU(3)$ Deconfining Phase Transition},
    to be published in Nucl Phys. {\bf B}.
   \bibitem{MAPAPE}
	G. Martinelli, G. Parisi and  R. Petronzio,
	Phys. Lett. {\bf B100} (1981) 485.
  \bibitem{SWESTA}
    R. H. Swendsen,
    Plenary talk given at the $18^{th}$ IUPAP International
    Conference on Statistical Physics (STATPHYS 18), Berlin, August 1992.
  \bibitem{KANNEL}
    Y. Kantor and D. Nelson,
    Phys. Rev. Lett. {\bf 58} (1987) 2774;
    Phys. Rev. {\bf A36} (1987) 4020.
  \bibitem{KAWNIM}
    H. Kawai and M. Ninomiya,
    Nucl. Phys. {\bf B336} (1990) 115.
  \bibitem{Migtalk}
    A. Migdal, talk given at Syracuse University, April 1992.
  \bibitem{Kostas}
    K. Anagnostopoulos, private communication.
  \bibitem{Colot}
    J. L. Colot,
    J. Phys. {\bf A16} (1983) 4423.
   \bibitem{BRDE}
	R. Brout and W. Deans, Nucl. Phys. {\bf B215}[FS 7] (1983) 407.
   \bibitem{BRDESI}
	R. Brout, W. Deans and A. Silovy, Phys. Rev. {\bf B27} (1983) 5813.
   \bibitem{ORBR}
	J. Orloff and R. Brout, Nucl. Phys. {\bf B270}[FS 16] (1986) 273.
  \bibitem{AMDUJO}
    J. Ambj{\o}rn, B. Durhuus and T. Jonnson,
    Nucl. Phys. {\bf B316} (1989) 526.
  \bibitem{AMDJ}
    J. Ambj{\o}rn, B. Durhuus J. Fr\"{o}lich and T. Jonnson,
    Nucl. Phys. {\bf B290} (1987) 480.
  \bibitem{WHEATE}
    R. Harnish and J. Wheater,
    Nucl. Phys. {\bf B350} (1991) 861.
  \bibitem{WHESTE}
    J. F. Wheater and P. W. Stephenson,
        Phys. Lett. {\bf B302} (1993) 447.
  \bibitem{AMBBP} J.M. Drouffe, G. Parisi and N. Sourlas,
    Nucl. Phys. {\bf B161} (1980) 397;
    J. Ambj{\o}rn, B. Durhuus and T. Jonnson,
    Phys. Lett. {\bf B244} (1990) 403.
   \bibitem{BALESH}
	W.A. Bardeen, B.W. Lee and R.E. Shrock,
	Phys. Rev. {\bf D14} (1976) 985.
  \bibitem{ShTob}
    S. Shenker and J. Tobochnik,
    Phys. Rev. {\bf B22} (1980) 4462.
  \bibitem{Brout}
    R. Brout, W. Deans and A. Silovy,
    Phys. Rev. {\bf B27} (1983) 5813.
  \bibitem{Brout2}
    R. Brout and W. Deans,
    Nucl. Phys. {\bf B215} (1983) 407;
    J. Orloff and R. Brout,
    Nucl. Phys. {\bf B270} (1986) 273.
  \bibitem{Apost}
    J. Apostolakis, C. Baillie and G. Fox,
    Phys. Rev. {\bf D43} (1991) 2687.
  \bibitem{Hasen}
    P. Hasenfratz and F. Niedermayer,
    Phys. Lett. {\bf B245} (1989) 522.
  \bibitem{winprog}
    M. Bowick, P. Coddington, L. Han, G. Harris and E. Marinari;
    work in progress.
  \bibitem{Butera}
    P. Butera, M. Comi and G. Marchesini,
    Nucl. Phys. {\bf B300} (1989) 1.
  \bibitem{LIPA}
    J.A. Lipa and T. C. P. Chui, Phys. Rev. Lett. {\bf 51} (1983) 2291.
  \bibitem{GOLBOOK}
    See N. Goldenfeld,
	{\em Lectures on Phase Transitions and the Renormalization Group },
	FIP Vol.~85 (Addison-Wesley, Reading Massachusetts, 1992).
  \bibitem{KAZIS}
    V. A. Kazakov,
        Phys. Lett. {\bf A119} (1986) 140.
  \bibitem{BOUKAZIS}
    D. V. Boulatov and V.A. Kazakov,
        Phys. Lett. {\bf B186} (1987) 379.
  \bibitem{simtemp}
    E. Marinari and G. Parisi,
    Europhys. Lett. {\bf 19} (1992) 451.
  \bibitem{MigLap}
    M. Agishtein, R. Benav, A. Migdal and S. Solomon,
    Mod. Phys. Lett. {\bf A6} (1991) 1115.
  \bibitem{GRPE}
	D. Gross and V. Periwal,
	Phys. Rev. Lett. {\bf 60} (1988) 2105.
  \bibitem{HOOFT}
	G.`t Hooft, 
	Nucl. Phys. {\bf B72} (1974) 461.
  \bibitem{BELA}
 	A. Belavin, A. Polyakov and A. Zamolodchikov, 
	Nucl. Phys. {\bf B241} (1984) 333.
  \bibitem{AAABJ}
	M. Agishtein and A. Migdal, 
	{\it Simulations of Four-Dimensional Simplicial Quantum
	Gravity}, Princeton preprint PUPT-1287 (1991)
	M. Agishtein and A.~Migdal, 
	{\it Critical Behavior of Dynamically Triangulated
	Quantum Gravity in Four Dimensions}, Princeton preprint PUPT-1311 (1992)
	J. Ambj{\o}rn and J. Jurkiewicz, 
	Phys. Lett. {\bf B278} (1992)42.
  \bibitem{HRICH}
 	W. Helfrich, 
	J. Phys. {\bf 46} (1985) 1263.
  \bibitem{ENZOP}
	E. Marinari and G. Parisi, 
	Phys. Lett. {\bf B240} (1990) 375.
  \bibitem{BOWICK}
	M.J. Bowick and E. Br\'ezin,
	Phys. Lett. {\bf B268} (1991) 21-28.
  \bibitem{PIBM}
	G. Fox, J. Apostolakis, M. Bowick, B. Br\"ugmann, P. Coddington, 
	D. Edelsohn, L. Han, G. Harris, E. Marinari, K. Mills and S. Ranka, 
	{\it Computer Science Software Tools and Languages to Support
	Computational Physics}, 
	report to the IBM Corporation.  
	SCCS-393.

  \bibitem{FEYINT}
	R. P. Feynman, 
	Rev. Mod. Phys. {\bf 20} (1948) 267.

  \bibitem{AVSP}
	Craig Upson, etc.
	IEEE Computer Graphics \& Applications. (1989) 30.

\end{thebibliography}

%\include{sec0}
% $Id: sec0.tex,v 1.4 1994/10/23 19:02:36 han Exp han $

\pagenumbering{roman}

\tableofcontents
\newpage

\newpage
\listoffigures

\newpage
\section*{Preface}
\addcontentsline{toc}{section}{Preface}

Thanks to 

\bigskip
\begin{flushright} Leping Han \\
Syracuse, October 1993
\end{flushright}

\newpage
\pagenumbering{arabic}

%\include{sec1}
% $Id: sec1.tex,v 1.6 1994/10/23 19:02:36 han Exp han $

\newpage

\section{Introduction}

In the twentieth century, two different formalisms have emerged which
can in principle, explain all known physical phenomena: general
relativity, which gives us a compelling description of the beginning
of the universe, and the quantum theory, which has unified the theory
of sub-atomic particles.  It seems the nature at its most fundamental
level, would have two sets of laws, based on two sets of principles,
two sets of equations, and operating on two different scales. It seems
hard to believe that nature could create a world split in such a
artificial fashion.

General relativity was historically derived from geometry. First came
Einstein's equivalence principle and the geometrical picture of
general covariance. From the geometry came a unique action, the
Einstein-Hilbert action. Then came the classical theory of curved
manifolds. Last comes the yet unfinished quantum theory of
gravitation. Thus, the historical sequence can be viewed as follows:

\begin{center}
$Geometry \rightarrow Action \rightarrow Classical \  Theory
\rightarrow Quantum \  Theory$
\end{center}

In late 70s and early 80s, the string theory has emerged as the
leading candidate for a theory of all known forces, including a finite
theory of quantum gravity.  However, string theory has its own
difficulties. The most striking feature of quantum string models is
perhaps their prediction of a critical space-time dimension outside of
which the quantum theory is problematical. For one of the most popular
models, the bosonic model, the critical space-time dimension is 26.

The bad feature of the bosonic model is the presence of a tachyon in
the spectrum, which violates causality. It is natural to seek
super-symmetric models with an equal number of bosonic and fermionic
variables. It is indeed a general rule of supersymmetry that
super-symmetric multiplets tend to cancel unwanted effects, which is
related to the zero point energy. Soon after Neveu and Schwarz found
another bosonic model realizing such a symmetry \cite{NeveuSchwarz}.
But this model is consistent (apart from the tachyons) still only in
specific critical dimensions of space time $d = 10$.  It is still far
away from our physical space-time 4 dimensional space.

People are more insterested in a string world-sheet with the standard
action embedded in a physically meaningful dimension. However, even
just for dealing with 4 dimensional space, we still have not known how
to define the path integral as used in quantum field theory. People
then started to think of using lattice formulations. Using Dynamical
Triangulated Random Surfaces is considered as a good discretized
version of the surfaces.  People would expext that once a phase
transition is found, a well-defined path-integral is obtained.  Thus,
the fomulation in the continuum limit is found. In doing so, a variety
of actions was proposed.

In our study, we present the results of a large-scale simulation of a
Dynamically Triangulated Random Surface with extrinsic curvature
embedded in three-dimensional flat space.  We measure a variety of
local observables and use a finite size scaling analysis to
characterize as much as possible the regime of crossover from crumpled
to smooth surfaces.  We also will discuss the computational approach
and the chanlenging issues involed in this study.

All works presented here were done in a collaborated manner by
computational physics group at Syracuse University. Myself
particularly have been devoted to developing random surface program,
optimizing program, taking data and analysing data and parallizing the
random surface program on different types of parallel computers.

First, I would like to briefly discuss several key fields and the
relations among them. These fields are string theory, quantum field
theory, quantum gravity, field theory and statistical mechanics and
Monte Carlo simulation. All of these fields play big role in modern
physics.  And I have been very excited and challenged by the facts
that random surfaces study is more or less related to different
aspects of all these fields.

\subsection{String Theory}

In classical theory, a point particle is described by a vector which
points from the origin to the location of the particle
$x_{\mu}(\tau)$, where $\tau$ parameterizes the location of the point
along the trajectory.  If the particle is a free relativistic particle
that moves in space-time, based on classical theory, its world line is
time-like and also has minimal proper length.

The appropriate action is

\beq
	S = -m\int ds
\eeq

where $m$ is the mass and $ds$ is the path length along the world
line. And the path length $ds$ can be written in terms of the
corrdinates:

\beq
	ds = -\sqrt{\dot{x}_{\mu}^2}d\tau
\eeq
where the dot refers to differentiation with respect to the parameter
$\tau$. 

One of the key properties of the action is its reparametrization
invariance, i.e. make a change of coordinates 
$\tau \rightarrow \tilde{\tau}(\tau)$, the action is invariant under
an arbitrary reparametrization of the variable $\tau$.

Similarly, a string is a one-dimensional extended object. The world
history of the string is a two-dimensional surfaces in space-time. The
natural generalization of the particle case is to postulate that a
free string (with free boundaries if it is open) is described by
surface with the following properties:

1. The surfaces is time-like, i.e., it possesses everywhere time-like
and space-like directions (except possibly at the boundaries).

2. It has extreme area, i.e., it is an ``extremum surface.''

Surprisingly, almost all of the main features of the Nambu-Goto string
have some forms of analogue in the first quantized point particle
theory. Many basic methods of quantization can be carried over
directly from the point particle case.(see \cite{BrinkHenn} ).

Let M be a fixed two dimensional manifold. Surfaces with the topology
of M are described by maps (we need a figure here to show the 
relation between tangent and surface and points). \cite{GreenGross}
\beq
	X: M \rightarrow {\bf R}^d.
\eeq
The parametric equations of the string world surface are 
\beq
	X^{\mu} = X^{\mu}(x^\alpha)
\eeq
where $\mu = 0,1, ..., d-1$ and $\alpha = 0,1$. We assume that
$x^{\alpha} \equiv (\tau, \sigma)$ provides a good parameterization. 

We will work in an Euclidean field theory in the following sections.

A string is described by a vector which points from the origin to its
location in space-time $X_{\mu}(\sigma,\tau)$. Its action is given by
the area swept out by the string as it moves in space-time 
(Fig. \ref{FIG1}).

One possible formulation for the action as the area of the world sheet was
introduced by Nambu, Goto and Hara. The area $A$ of the surface swept
out by the string is given by
\beq
	A = \int_{M}d^2x\sqrt{detg_{ij}}
\eeq
where
\beq
	g_{ij} = {\frac{\partial{X^{\mu}}}{\partial{x^{i}}}}
			      {\frac{\partial{X^{\mu}}}{\partial{x^{j}}}}
\eeq
is the induced metric on the surface embedded in d-dimensional space
\beq
	ds^2 = dX^{\mu}dX^{\mu} =
	{\frac{\partial{X^{\mu}}}{\partial{x^{i}}}}
	{\frac{\partial{X^{\mu}}}{\partial{x^{j}}}} dx^idx^j
\eeq
The action of the free string is proportional to the area and hence
given by Nambu-Goto action \cite{NambuGoto}
\beq
\protect\label{action1}
	S\left[X^{\mu}(x^{\alpha})\right] =
-{\frac{1}{2\pi\alpha^{'}}}{\int^{\tau_2}_{\tau_1}d\tau}
{\int^{\pi(open) or 2\pi(closed)}_0d\sigma \sqrt{-detg}}
\eeq
where $\alpha^{'}$ is a constant with dimension of length-squared in
units of $\hbar$. 

The above action (\ref{action1}) has one drawback: it is not
quadratic in the fields.  This is a serious problem if one wants to
quantize the string by means of path-integral methods, for the usual
Lagrangian form of the path integral is only valid for quadratic
actions.

Polyakov \cite{POLSTR} proposed that in the quantum theory one replace
the area action by an equivalent classical action that depends on an
additional intrinsic metric $g_{ij}$ (continuum model). The new action
is a functional of both the intrinsic metric $g_{ij}$ and the
coordinates $X$ with the quadratic form:
\beq
\protect\label{polyfor1}
	S[X,g] = \int_{M} d^2x \frac{1}{2} \sqrt{detg} g^{ij}
		 {\frac{\partial{X^{\mu}}}{\partial{x^{i}}}}
		 {\frac{\partial{X^{\mu}}}{\partial{x^{j}}}}.        
\eeq

Classically, one can eliminates $g_{ij}$ from the equation
(\ref{polyfor1}) algebraically and substituting this solution back to
the action. The resulting expression is the Nambu-Goto action, namely,
the area of the world sheet (definitely a reparametrization-invariant
expression).  Quantum mechanically the limination of $g$ involves
performing a path integral\cite{BrinkHenn}. In general an extra
``Liouville mode'' is left over (the Weyl invariance is broken) except
in the special case of d = 26. There are many leteratures to show the
ways to derive the critical dimension, i.e., $d = 26$. 

\subsection{Quantum Field Theory}

Any quantum theory has to be understood on the level of quantum field
theory. So is string theory. We can think string theory as a field
theory by direct analogy with ordinary quantum field theory. The basic
object of ordinary quantum field theory if a field $\phi(x^{\mu})$
where $x^{\mu}$ is a coordinate on space-time. The dynamics of this
field can be described by an action $S[\phi]$. Given $S[\phi]$ one may
then compute, e.g., the amplitude $A(x^{\mu},y^{\mu})$ for a particle
observed at $x^{\mu}$ to appear at $y^{\mu}$ via functional
integration:
\beq
	A(x^{\mu},y^{\mu}) = \int {\cal D}\phi^{iS[\phi]}
			    \phi(x^{\mu})\phi(y^{\mu})
\eeq
In string theory, particles are replaced by strings. While the
location of a particle is described by a single coordinate $x^{\mu}$,
the location of a string is described by a function $x^{\mu}(\sigma)$,
where $\sigma$ is a parameter along the string. One then wished to
compute. e.g., the amplitude $A[x^{\mu}(\sigma),y^{\mu}(\sigma)]$ for
a string observed at $x^{\mu}(\sigma)$ to appear at $y^{\mu}(\sigma)$.
This requires a string field $\phi[x^{\mu}(\sigma)]$ whose domain is
the space of all configurations $x^{\mu}(\sigma)$. Given some action
$S[\phi]$, the amplitude may then be computed just as in ordinary
quantum field theory:

\beq
	A[x^{\mu}(\sigma),y^{\mu}(\sigma)] = \int {\cal D} \phi e^{iS[\phi]}
			  \phi[x^{\mu}(\sigma)] \phi[y^{\mu}(\sigma)] 
\eeq

In summary, string field theory is quantum field theory with
space-time replaced by string space; i.e., the space of all string
configurations.

But the key question again in string filed theory is to find a
suitable action $S[\phi]$. To construct the action, what kinds of
properties the action $S[\phi]$ should have?

The principle of renormalizability tells us that the action should be
constructed out of covariant expressions in $X^{\mu}$ and $g_{ij}$
which are of dimensions two or less and which respect the Euclidean
invariance of physical space-time. The most general such action
consists of three terms\cite{GreenGross}:
\beq
	S[X^{\mu},g_{ij}] = A\frac{1}{2}\int_Md^2x\sqrt{g}g^{ij}
	\partial_iX^{\mu}\partial_jX^{\mu} +
	B\int_Md^2x\sqrt{g} + C\frac{1}{4\pi}\int_Md^2x\sqrt{g}R
\eeq

The first term in the above is just the classical action we had
previously discussed. The second term is a cosmological constant. The
third term is a topological invariant called the Euler characteristic
$\chi$:
\beq
	\chi(M) = \frac{1}{4\pi}\int_Md^2x\sqrt{g}R.
\eeq
The Euler characteristic $\chi(M)$ is related to the topology of the
manifold via the relation
\beq
	\chi(M) = 2 - 2h -b,
\eeq
where $h$ is the number of handles on the surface and b the number of
boundaries.

\subsection{Field Theory And Statistical Mechanics}

The connection between field theory and statistical mechanics is best
understood using the Feynman path integral approach\cite{FEYINT}.

A standard computation in quantum mechanics is the probability
amplitude for a transition of a point particle from the initial state
$q_i$ at time $t_i$ to the final $q_f$ at the time $t_f$. It is called
propagator, and can be expressed in Heisenberg picture as
\beq
	K(q_ft_f;q_it_i) = \langle q_ft_f|q_it_i \rangle 
\eeq
The feynman formulation of quantum mechanics states that
\beq
	K(q_ft_f;q_it_i) = \sum_{all \  possible \  paths} e^{iS/\hbar}
\eeq
where S is the classical action for a path $x(t)$ connecting the
initial point to the final point(figure). The sum over all possible
paths is generally written as a ``path integral'' (or functional
integral)

\beq
	K(q_ft_f;q_it_i) = \int[{\cal D} x(t)] e^{iS/\hbar}
\eeq

In the case of quantum field theory, the fundamental objects are not
paths but fields $\Phi(x)$, and the propagators are vacuum expectation
values of time ordered products of the fields. In the Feynman
formulation they are expressed as
\beq
	\langle \Phi(x_1)\Phi(x_2) \rangle \equiv \langle 0|T[\Phi(x_1)\Phi(x_2)]|0\rangle
\eeq
\beq
\protect\label{vacgen}
	\langle \Phi(x_1)\Phi(x_2) \rangle = \int[{\cal D} \Phi(x)] \Phi(x_1)\Phi(x_2) e^{iS/\hbar}
\eeq
where  the action is the space-time integral of the Lagrangian
\beq
	S(\Phi(x)) = \int d^4x{\cal L}(\Phi(x))
\eeq
The functional integral represents the sum over all psossible field
configurations $\Phi(x)$. The vacuum expectation values of the above
equation (\ref{vacgen}) can be obtained from a ``vacuum generating
functional''
\beq
	Z = \int[{\cal D} \Phi(x)] e^{iS/\hbar}
\eeq
in a similar manner to the calculation of correlation functions and
other thermodynamic variable from the partition function

\beq
	Z = \sum_s e^{-H(s)/kT}
\eeq
in statistical mechanics. Here $H$ is the Hamiltonian, $kT$ is the
Boltzmann constant times the temperature, and the sum runs over all
psossible states of the system.

We wish to convert the above $Z$ to a form similar to the partition
function in statistical mechanics. We change variables to imaginary
time (called the Euclidean time in quantum field theory), i.e. perform
a Wick rotation $t \ \rightarrow \ -it$, the Euclidean action $S_E =
-iS$ and the function $Z$ can be expressed as
\beq
\protect\label{pathint}
	Z = \int[{\cal D} \Phi(x)] e^{-S_E/\hbar}
\eeq
If we indentify $S_E/\hbar$ with $H/kT$, the partition function of
statistical mechanics and the generating functional of Euclidena
quantum field theory are formally identical.  The result provides the
link between quantum field theory and statistical mechanics.  The
$\hbar$ appears here because we are dealing with quantum mechanics
system. In classical mechanics the actual path of the particle is
determined by finding an extremum for the action S. In quantum
mechanics all paths contribute to the probability amplitude with a
weight $e^{iS/\hbar}$.  And $\hbar$ sets the scale of quantum
fluctuations just as the temperature sets the scale for thermal
fluctuations. Correlation functions in statistical mechanics are
equivalent to Euclidean space propagatoir in field theory. 

The equation (\ref{pathint}) is purely formal until a workable
definition is given to the functional integral. For example, a free
particle's path integral reads
\beq
	\Sigma \rightarrow \int{\cal D} x = \lim_{N \  \rightarrow \
\infty} \int {\prod}^{3}_{i=1} {\prod}^{N}_{n=1} dx_{i,n}
\eeq
where the index $n$ labels $N$ intermediate points that divide the
interval between the initial and the final coordinates. We will take
the limit when N approaches infinity. We must also be able to
regularize all the usual divergences. The lattice approach solves both
these problems. On a finite lattice, the finctional integral is
approximated by a well defined finite dimensional integral, and
(\ref{pathint}) is obtained bu taking the contimuun limit, in which
the lattice spacing goes to zero while the lattice volume tends to
infinity. And it can be shown that the correlation length of a lattice
field theory is related to the mass gap by $\xi \ \sim \ m^{-1}$.

Having properly defined the Euclidean space generating functional and
seen that it is mathematically equivalent to a partition function,
powerful techniques from statistacal mechanics such as low and high
temperature expansions, mean field theory, renormalization group
analysis and Monte Carlo simulation may be applied to the study of
quantum filed theory. Such techniques are especially useful when the
theory is discretized onto a lattice.

Over the past 60 years, many predictions of the standard model,
including predictions from QED and QCD, have been confirmed to a high
degree of accuracy by accelerator experiments. Exact solutions of the
QFT equations do not exist at this time. In most cases the predictions
of the standard model have been obtained using perturbation theory
because electro-magnetic, weak, and strong interactions at high
energies involve small values of the coupling constant. Small values
are especially applicable for scattering experiments involving
electrons or protons at very high energies of 100 Gev and more.
However, other phenomena such as nuclear structure, proton structure,
and nuclear matter at high coupling and therefore prohibit the use of
perturbation theory. One of the most important challenges facing
quantum field theorist today is to solve QFT in the nonperturbative
regime. A potential solution to this challenge was found in the late
1970s with the development of a formalism called lattice field theory
(LFT). In its most useful formulation, LFT makes much use of numerical
techniques, particalrly Monte Carlo methods familiar in
condensed-matter physics. The application of LFT to the strong
interactions has required hundreds and sometimes thousands of hours of
supercomputer time. LFT has helped spur the development of
supercomputers-especially parallel architectures, as LFT requires only
local interactions.

With the light of LFT and as we have seen, the basic feature of string
theories is the geometric origin of the action of the free bosonic
string, the the Nambu-Goto action. It was constructed as the area of
the world surface traced out by the propagating string.

The path integral approach to quantized strings postulates that the
partition function is given by
\beq
	Z = \sum_{surfaces}exp(-area).
\eeq
In the above equation one is supposed to sum over surfaces with all
topologies.

\subsection{Quantum Gravity}

As we mentioned above, one of the key motivations of studying string
theory is to find a theory that contains quantum gravity. Quantum
Field Theory and the theory of General Relativity are, separately,
probably the two most successful physical theories of this century.
Nobody has yet been able to bring the two together into one complete
and consistent quantum theory of gravity.

Thus, the construction of a theory of quantum gravity which
encompasses both General Relativity(GR), or some of its extensions,
and the principles of Quantum Mechanics is probably one of the most
difficult and exciting problems of theoretical physics. It is
well-known that there are at least two basic difficulties. Firstly,
One major impediment to such a theory is that, unlike gauge field
theories, gravity with the Einstein-Hilbert action
\beq
    S = \frac{1}{16\pi G_N}{\int d^4x{\sqrt{g}}R}
\eeq
is not renormalizable, at least not by the usual methods of
perturbation theory.  Secondly, the Einstein equations (the evolution
equations for the metric) are derived from an action principle, but
the Einstein-Hilbert action is unbounded from below, and a naive path
integral formulation of a quantum theory is expected to be
mathematically ill defined.  Among a list of attempts to define
quantum gravity including canonical quantum gravity, standard field
theory using perturbation theory, quantum field theory in curved space
times, the path integral approach to Euclidean quantum gravity,
quantum cosmology, discrete models of quantum gravity, and topological
field theory.  There is not a single proposal for a quantum theory of
gravity that is self-consistent as a physical theory, even allowing
for a great deal of incompleteness. If one examines the theories
closely, they are not flawless to begin with\cite{Bruegman}.

Although quantum field theory and general relativity seem totally
incompatible, the past two decades of intense theoretical reseach have
made it increasingly clear that the secret to this mystery most lies
in the power of guage symmetry. At present, the most promising hope
for a truely unified and finite description of these two fundamental
theories is the superstring theory. Superstrings possess by far the
largest set of gauge symmetries ever found in physics. This has lead a
number of physicists to adopt the position that General Relativity is
only the low energy limit of some other quantum theory, such as
superstring theory. To pursue a quantum theory of gravity, one needs a
nonperturbative method of calculation: the methods of lattice field
theory, which have already been applied to gauge theories, are
immediately suggested.

In the lattice formulations of quantum gravity, one tries to construct
a discretization procedure which makes the functional integration over
metrics (or over the other degrees of freedom which describe classical
gravity) meaningful, by truncating this infinite dimensional integral
into a sum over a finite number of variables. Then one should look for
critical values of the parameters (the coupling constants) of the
discretized theory where a scaling behavior occurs, so that the
details of the discretization become irrelevant and a continuum theory
can be constructed. This lattice approach has been much successful for
studying ordinary relativistic quantum field theories, such as
non-Abelian gauge theories, and in some cases to construct them
rigorously. When trying to apply these ideas to General Relativity one
hopes that a quantum theory can be obtained from a path integral
formulation based on the Lagrangian formulation of General
Relativity, namely from the Einstein-Hilbert action, which is a
functional of the metric tensor $g$, taken as the dynamical variable.

Therefore, we are lead to consider the functional integral over
Euclidean metric
\beq
	\int D[g] e^{S_E[g]}
\eeq
with the Euclidean Einstein\-Hilbert action
\beq
	S_E[g] = \frac{1}{16 \pi G_N} \int d^Dx \sqrt{|g|}(-R + \Lambda)
\eeq
($D$ is the dimension of space-time, $G_N$ Newton's constant and
$\Lambda$ the cosmological constant.

\subsection{Random Surfaces}

More recently the connection between random surfaces, quantum gravity
and string theory has been the subject of many investigations.  In
statistical mechanics and condensed matter physics problems involving
interfaces (for instance wetting) or two dimensional defects (for
instance charge density waves) as well as two dimensional films or
membranes (mono-and-bi-layers, cell membranes) are naturally
formulated in terms of fluctuating surfaces.

In string theory, the worldsheet, however, must fluctuate as one is
required to integrate over all admissible metrics to enforce
diffeomorphism (reparametrization) invariance. In this way new
intrinsic degrees of freedom (the conformal modes of the metric) enter
the theory. From the statistical mechanics viewpoint one is thus
dealing with an exciting class of models described by certain order
fields living on a fluctuating substrate. Averaging over metrics
corresponds to being in the universality class of translationally and
orientationally disordered fluctuating surfaces or membranes. These
are often called liquid-like membranes, as opposed to crystalline or
hexatic membranes that are translationally or orientationally ordered
respectively\cite{NELSON}. The remarkable fact is that these
statistical mechanical models defined on a random mesh are, in a
sense, easier to solve than the conventional models defined on a rigid
regular lattice. This is because diffeomorphism invariance reduces the
number of effective degrees of freedom. It is even possible to admit
fluctuations which change the topology of the surface (growth or
collapse of handles).

This is certainly of great interest as a model of gravity but also
provides the basis for an exploration of membranes. Recently
particularly simple models corresponding to certain types of conformal
matter coupled to $2d$-gravity have been exactly solved including the
sum over all possible topologies. The outline of the solution together
with the relation to $2d$-gravity and extensions to more realistic
random surfaces (flexible liquid-like membranes) is discussed
below(fugure).

An exciting development took place during the last two years: the
idea, advocated by various authors, that random lattices may provide a
sensible discretization of two dimensional Euclidean quantum gravity
was confirmed quantitatively. Exact results for the continuum theory,
obtained by conformal fields theory methods, are in agreement with
exact solutions for the discretized models.)

If we only use Nambu-Goto action as our string, it is well known that
there is Tychon at the ground state. Only we can get rid of the Tychon
in 26 Dimension. 

In order to perform the functional integral, we have to define the
integral on a lattice. We know that if there is a phase transition at
a certain point, we can say that this point is the fixed point, and we
can make a continuum limit around this point, since the details of the
discretization is not important at all.

In the search for a discretized version of a bosonic string theory,
the model of triangulated random surfaces with Gaussian action was
proposed in 1985 \cite{DYTRAS}. It was hoped that this formulation
could be used to understand the properties of a bosonic string in
sub-critical dimensions $d < 26$. Investigations carried out over the
past few years seem, however, to indicate that for any $d > 1$, it
degenerates into a theory of branched polymers. Lattice approach also
show the surfaces are crumple.

To overcome this problem it was suggested to add to the action a new
term, proportional to the extrinsic curvature, which would smoothen
the surface and possibly avoid the collapse into a branched polymer.
The importance of the extrinsic curvature for the description of
surfaces was pointed out many years ago in condensed matter and
biophysics. On a triangulated surface various forms of this term have
been investigated. 

In spite of many new ideas and exciting development in random
surfaces, we have to admit that we have not established the very clear
relationship to the continuum theory from this approach.  Here we are
be concerned with an attempt to regularize the formal euclidean path
integral for quantum gravity by a discretization of the integration
over the space of metrics. Such a regularization may lead to a well
defined theory. However it may be completely unrelated to the
'correct' quantum theory of gravity in 4 dimensinal real space. The
relevant question to ask is therefore whether the such a regularized path
integral leads to reasonable physical statements. A much more modest
goal is to understand the statistical model resulting from such a
discretization, and this is what we want to contribute to.

\subsection{Monte Carlo Simulation}

The Monte Carlo evaluation of the path-integral (\ref{pathint}) is
largely standard. One constructs a Markov chain $\{T_i\}$ of
triangulations with equilibrium configuration exp$(-S_E)$ so that
\beq 
    \langle f \rangle = \lim_{n\rightarrow\infty} {1\over n}
    \sum_{i=1}^n f(T_i).
\eeq

Of interest is the phase diagram obtained for different coupling
constant. 

Given the set of ergodic moves, we have to define a random walk
through the space of triangulations and compute the weights
required for detailed balance. Any algorithm to pick a move will
prescribe a probability $P(T\rightarrow T')$ for a transition from
triangulation $T$ to $T'$ that is not independent of the
triangulations involved since the triangulation, and therefore for
example the number of allowed moves, is changing.  In general, suppose
the transition probability is
\beq
    W(T\rightarrow T') = P(T\rightarrow T')W_M(T\rightarrow T'),
\eeq
where $W_M$ is the standard Metropolis weight. Detailed balance for
$W$ implies
\beq
    P(T\rightarrow T') = P(T' \rightarrow T),
\eeq
since $W_M$ already satisfies detailed balance. 

more to write

\subsection{Summary}

The bosonic string for surfaces embedded in $26$ dimensional space has
been studied extensively.  Much progress in numerically simulating
strings has been made through the use of Dynamically Triangulated
Random Surfaces (DTRS)\cite{DYTRAS,KAZMIG}.  For theories of surfaces
embedded in $ D \leq 1$, analysis of the continuum Liouville theory
\cite{KPZ,DADIKA} and of matrix models has led to consistent
predictions for critical exponents and correlation functions.  In the
`double scaling limit', in fact, exact nonperturbative solutions
\cite{EXACT} have been found from these matrix models; the functional
integral over surfaces has been essentially summed over all genera.

These analytical techniques have failed, for the most part, in probing
the theory of random surfaces in the more interesting embedding regime
$D > 1$.  Indeed the formulas for critical exponents computed in
Liouville theory give complex results when continued to the range
$1<D<25$.\footnote{More precisely, one encounters these instabilities
in Liouville theory when the quantity $c - 24\Delta >1$, where $c$
denotes the central charge of the matter theory which describes the
embedding of the surfaces and $\Delta$ is the conformal weight of the
lowest weight state in this theory \cite{KUTSEI}.  Since here we are
considering flat space, $c=D$ and $\Delta=0$.} The matrix models
describing $D>1$ strings have so far been too complicated to solve.

Monte Carlo simulations for $D>1$
\cite{BILDAV,AMNOEX,EALRYM,MINOEX} indicate that these theories do
not appear to describe the fluctuations of two dimensional {\em smooth}
surfaces in the continuum limit.  Extremely spiky, branched-polymer-like
configurations with high (perhaps infinite) Hausdorff dimension dominate the
functional integral\footnote{In the same way singular configurations
dominate the Gaussian theory, which is essentially a theory of free random
walks.}.

Evidence for this pathology was obtained, for instance, when it was
shown that the bare string tension, essentially the amount of work per
unit area needed to perturb a boundary loop on these configurations,
cannot vanish at the critical point \cite{AMNEED}. (This implies that
the renormalized string tension diverges in the continuum limit.) It
has been speculated in ref. \cite{CATES} that the proliferation of
vortex configurations of the internal geometry (the `Liouville mode')
induces the degeneration of these surfaces in the embedding space.  A
related explanation that is often proffered is that a negative
mass-squared particle\footnote{In some contexts this is referred to as
the `tachyon'.}, which comes on shell in the string theory for $D >1$,
creates instabilities which are made manifest by these singular
configurations.

The tachyon, and apparently these related instabilities, can be eliminated in
particular cases by introducing fermionic coordinates and supersymmetry on
the worldsheet, and implementing an appropriate projection of states.
Presumably, the fermions effectively smooth out the
surfaces.  This would be consistent with what has been observed for
one-dimensional geometries; the random walk of a spin one-half particle has
Hausdorff dimension one and thus appears to be smooth \cite{POLBOO}. Many
authors have proposed an alternative modification of the string action
\cite{CANHAM,HELFRI,POLEXT,KLEINA} via the addition of a term that directly
suppresses extrinsic curvature\footnote{In fact, integrating the fermions out
of the Green-Schwarz superstring yields an action similar to the one we
consider, but with the addition of a complex Wess-Zumino type term
\cite{WIEGMA}}. 

%\include{sec2}
% $Id: sec2.tex,v 1.9 1994/10/23 19:02:36 han Exp han $

\newpage

\section{More About Random Surfaces analytically and numerically}

As we have discussed in previous session, string theory, has been
conjectured to describe the underlying fundamental physics of a wide
variety of physical phenomena and models. These include the strong
interaction at long distances, the three-dimensional Ising model and
unified models incorporating gravity.  In its simplest form, the
bosonic string, it is a theory of free fluctuating surfaces.  The
functional integral for the Euclideanized bosonic string is just the
partition function for an ensemble of random fluctuating fluid
surfaces.  Such surfaces are also ubiquitous in nature, being found
for example in macro-emulsions and the lipid bilayers that form an
important part of cell membranes
\cite{DAVREV}.  These systems are fluid because their component
`molecules' are loosely bound. Their constituents are arranged so that
the net surface tension (nearly) vanishes; thus these membranes are
subject to large thermal fluctuations.  In one important respect,
however, these chemical/biological membranes differ fundamentally from
the surfaces we discuss and simulate; they are self-avoiding.  The
world-sheets of the bosonic string, in contrast, generically
self-intersect.

The theory of $2d$ fluid random surfaces embedded in $R^3$, with an
extrinsic curvature term (bending rigidity) in the action, has
received considerable analytical and numerical attention in the last
decade \cite{DAVREV,AMBTWO,BCHHM}.  

\subsection{2D Gavity and Random Surfaces}

\subsubsection{The Motivation of Random Surfaces -- Continuum Model}

Let us start by considering only a flucuating 2d-surface with no
matter (order fields) at all. This is clearly two-dimensional gravity.
Since there are no embedding string coordinates it is also a model of
strings in zero dimensions.  The Einstein-Hilbert action with a
cosmological constant term for $2d$ gravity is
\beq
\protect\label{EH}
	S[g] = \frac{-1}{16 \pi G} \int_\Sigma d^2 \xi \sqrt{g} R
     + \mu \int_\Sigma d^2 \xi \sqrt{g}
\eeq
where $g_{\alpha \beta}(\xi_1,\xi_2)$ is the $2d$ metric of the Riemann
surface $\Sigma$ with coordinates $\xi_1$ and $\xi_2$.
These strings sweep out two-dimensional Riemann surfaces as they
evolve in Euclidean time. In the first quantized description of string
theory one may view the string coordinates describing the embedding of
the worldsheet in the target spacetime as a collection of scalar
fields living on the worldsheet.

The partition function $Z$ then depends on two variables, Newton's
constant $G$ and the cosmological constant $\mu$
\beq
	Z[G,\mu] = \int\left[{\cal D}g\right]\  e^{-S\left[g\right]}
\eeq
where the path integral is over all admissible metrics of Riemann
surfaces $\Sigma$. In two dimensions the action (\ref{EH}) is simple since
the first term is a topological invariant by the Gauss-Bonnet theorem
\beq
	S= \frac{-\chi(\Sigma)}{4G} + \mu A(\Sigma)
\eeq
where $\chi$ is the Euler characteristic of $\Sigma$ and $A$ is the
area. $\chi$ is related to the number of handles, or genus $h$, by
$\chi = 2 - 2 h$, where for simplicity we are assuming $\Sigma$ to be
closed (without boundaries). The partition function thus reduces to
\beq
\protect\label{parttwo}
	Z\left[G,\mu\right]=\sum_h \int dA e^{\frac{\chi}{4G}} e^{-\mu A}\Omega_h(A)
\eeq
where $\Omega_h(A)$ is the density of states of Riemann surfaces
$\Sigma$ of fixed area $A$ and genus $h$,
\beq
	\Omega_h(A) \equiv \int_{(h;A)} {\cal D} g_{\alpha \beta}
\eeq
$\Omega_h(A)$ is very difficult to calculate as $h$ increases and the
sum over genus in (\ref{parttwo}) diverges \cite{GRPE}.

The above expressions are all, in fact, ill-defined. To give them
meaning we must regularize the path integrals. 

\subsubsection{The Discretization of Random Surfaces -- Dynamically
Triangulated Random Surfaces}

We shall be concerned primarily with the Polyakov form of the string
action \cite{POLSTR}, in which an additional {\em intrinsic metric}
$g_{ij}$ is introduced to describe the surface geometry.  

One approach is to discretize by replacing $\Sigma$ by a lattice. A
particularly concrete and appealing discretization is to consider all
triangulations (or more generally cellular decompositions) of
$\Sigma$. The surface is thus replaced by a discrete set of $n$ points
(vertices) labelled by an index $i$. The connectivity of the lattice
is described by the adjacency matrix
\beq
 C_{ij} = \left\{ \begin{array}{cc}
       1 & \mbox{ if $i$ and $j$ are connected by a link} \\
       0 & \mbox{ otherwise }
       \end{array}
       \right.
\eeq
This defines a metric on the lattice by fixing all links to have
length one in intrinsic metric. Thus all triangles (cells) in the
triangulation are equilateral and of fixed area. 

In the case, the coordination number at each vertex determines the
intrinsic curvature of the surface.  The coordinates $i$ label the
vertices of the triangulation. Then the discrete analogue of the
intrinsic metric is the adjacency matrix $C_{ij}$ whose elements equal
$1$ if $i$ and $j$ label neighboring nodes of the triangulation, and
vanish otherwise.  Two-dimensional diffeomorphism invariance reduces
to the permutation symmetry of the adjacency matrix at this discrete
level.  One of the keys, in fact, to the power of this construction is
the preservation of this symmetry. Each vertex of the triangulation is
embedded in $R^3$ via the mapping $X^{\mu}_i$.  

Given the embedding $X$, we can also associate a unit normal vector
$(n^{\mu})_{\hat{k}} $ with each triangle on the surface (Roman
indices with hats label the triangles).  Note that all of the surface
curvature of our triangulations is concentrated along the links and
vertices. The surface is still flat in the direction tangent (but not
transverse) to each link, so that the mean curvature has support on
the links, while the Gaussian curvature is non-zero only at the
vertices.  

The Euler characteristic follows from Euler's relation $\chi=V-E+F$,
for $V$ vertices, $E$ edges (links) and $F$ faces (triangles). Local
curvature(intrinsic curvature) is defined by means of the deficit
angle
\beq
	R_i = \pi\frac{(6 - q_i)}{q_i},
\eeq
where $q_i$ denotes the connectivity of the lattice at vertex $i$.
\beq
 q_i = \sum_{j} C_{ij}
\eeq
The Gaussian curvature $K$ on the other hand is expressed in terms of
the deficit angle in the embedding space.

To simulate the integral over metrics the adjacency matrix must be
allowed to fluctuate so that the coordination number of a node becomes
a dynamical degree of freedom. The local environment of a node is
constantly changing.  This considerably complicates the study of such
models from a computational point of view but also makes them more
interesting. These models are called Dynamically Triangulated Random
Surfaces DTRS \cite{DYTRAS}. The basic move to update $C_{ij}$ is a
flip on a fundamental parallelogram of two triangles sharing a common
edge(figure flip-move).

The discrete version of the partition function (\ref{parttwo}) replaces
integrals over metrics by sums over admissible triangulations and may
be written in the form
\beq
\protect\label{disc}
Z[G,\mu]=\sum_{h=0}^{\infty} e^{ {2-2h} \over {4G} }
\sum_{n=0}^{\infty} e^{-\mu n} Z_{h,n}
\eeq
where $Z_{h,n}$ is the number of distinct triangulations with $n$ vertices
and genus $h$. $Z_{h,n}$ is a discrete version of $\Omega_h(A)$, since
$A$ is proportional to the $n$ for fixed area elementary triangles.

\subsubsection{The Problem of This Model of Random Surfaces}
It has been shown that with including particular kinds of matter
living on the surface, the string susceptibility can be expressed
as 
\beq
\protect\label{kpz}
	\gamma_h = 2 - \frac{(1-h)}{12} \left\{ 25-c + \sqrt{(1-c)(25-c)}\right\}
\eeq
where $c$ is called central charge, whcih measures the response of the
free energy to the loacl curvature of freedom of the model. If $c \ =
\  1$, the mean surface area
\beq
\langle A \rangle = -\frac{\partial \log Z}{\partial \mu}
\eeq
diverges as ${1 \over {\mu - \mu_c}}$.
where  $\mu_c$ is the critical cosmological constant.

The string susceptibility $\gamma_h$ is clearly the critical exponent
for the specific heat. Diverging surface area is an indication of
criticality.  Near $\mu_c$ one may thus construct a continuum limit
with associated critical exponents that are universal in the sense
that they do not depend on the fine details of the lattice. 

Suppose now that we wish to describe more realistic string models
corresponding to surfaces embedded in a target space of dimensionality
$d$ greater than one. The surface is given by $x^{\mu}(\xi_1,\xi_2)
\  (\mu=1,...,d)$. These models obviously have $c > 1$. An immediate
problem is then apparent from equation (\ref{kpz}). According to the
continuum results the string susceptibility is imaginary for $1<c<25$.
This suggests that the model has an inherent instability. 

Analytical and computational investigations clearly establish that the
continuum limit of these models is dominated by surfaces which
degenerate into a branched tree of tubes of diameter of order the
lattice spacing. These are called branched polymer configurations and
are more one-dimensional than two-dimensional. The origin of these
spikes is clear in the Nambu-Goto formulation since an infinitesimally
thin long tube has vanishing area and is therefore not suppressed by
the area action. The large entropy for such configurations eventually
dominates the statistical mechanics of these surfaces. The Polyakov
action has been shown to be in the same universality class.

\subsection{Extrinsic Curvature -- A Possible Additional Terms to the Initial Model of Random Surfaces}

\subsubsection{Extrinsic Curvature -- How to Construct It}


Since the tubes of the branched polymers have a high extrinsic
curvature it is possible to suppress these (and other irregular)
configurations by adding a bending rigidity term to the action
\cite{POLEXT,KLEINA,HRICH}.

To write down our action we introduce an explicit parametrization of a
generic surface $\cal M$ in $R^{3}$ with coordinates $(\sigma _1,
\sigma _2)$ and the embedding $X^{\mu}(\sigma _i)$. $\mu$ runs from
$1$ to $3$ (since we only study the case of a $3d$ embedding space).
The induced metric (the pullback of the Euclidean $R^{3}$ metric via
the embedding) is given by
\beq
  h_{ij} = \partial _{\sigma_i} X^{\mu}\partial _{\sigma_j} X_{\mu}\ .
\eeq
We will use Greek letters for the embedding space indices; they can be
raised and lowered at will since our background space is flat.
Associated with each point in $\cal M$ are tangent vectors ($t^{\mu}_i
{\in} T\cal M$) and a normal vector $n^{\mu} {\in} T\cal M^{\perp}$.
The extrinsic curvature matrix $K_{ij}$ (the second fundamental form)
can be defined by
\beq
\protect\label{kijdef}
	 \partial_{i} n^{\mu} = - K_{ij}{t^{\mu}}^j \ .
\eeq
The eigenvalues of this matrix are the inverses of the radii of
curvature of $\cal M$. One usually describes the geometry of these
surfaces in terms of the mean curvature \cite{CHOBRU,DOCARM}
\beq
	    H = \frac{1}{2}h^{ij}K_{ij}\ ,
\eeq
and the Gaussian curvature
\beq
	   K = \epsilon^{ik}\epsilon^{jl}K_{ij}K_{kl}\ .
\eeq
where the extrinsic curvature matrix (Gauss' second fundamental
form) is $K_{ij}$ and given by
\beq
	K_{ij}^{\mu} = D_iD_jx^{\mu} 
\eeq
where $D_i$ is the covariant derivative along the surface. This is the
only additional term relevant under rescaling $x \rightarrow \lambda x$
that may be added to the string action and so will eventually be generated
by radiative corrections in any case. 

One can show that the Gaussian curvature can be computed solely from
the metric $h_{ij}$, while the mean curvature depends explicitly on
the embedding $X^{\mu}$. In three dimensions the trace of $K$ is the
mean curvature $H=1/r_1 + 1/r_2$, where $r_i$ are the principal radii
of curvature of the surface. 

The extrinsic curvature action is
\beq
	S_{EC} = \kappa \int d^2\xi \sqrt{g} ({\rm Tr} K)^2 
\eeq
Its discrete form may be written as
\beq
S_{EC} = \kappa \sum_{<ij>} ( 1 - {\bf \hat n}_i\cdot{\bf \hat n}_j)\quad
\eeq
where $i$ and $j$ represent triangles that share a common edge and
${\bf \hat n}_i$
is the unit normal to triangle $i$. 

\subsubsection{Perspective -- with Extrinsic Curvature}

$S_{EC}$ clearly suppresses local fluctuations in the mean curvature
of the surface. But the key question is whether there is long-range
order in the normals to the surface. The bending rigidity is, in fact,
a running coupling --- it depends on the scale at which it is
measured. A perturbative calculation in the inverse coupling
$\kappa^{-1}$ reveals that strings with bending rigidity are
asymptotically free in the same sense as Quantum Chromodynamics.
Fluctuations screen the theory and soften the effective bending
rigidity as the length scale increases. The momentum $p$ dependence of
$\kappa$ is found to be \cite{POLEXT}
\beq
\kappa^{-1}(p) = \frac{\kappa^{-1}_0}{1 - \frac{d}{2}
\frac{\kappa^{-1}_0}{2 \pi} {\rm log} \frac{\Lambda}{p}}
\eeq
where $\Lambda$ is the cutoff or inverse lattice spacing and $d$ is the
dimensionality of the target space(figure for the coupling).

At large length scales $\kappa$ tends to zero and there is no suppression of
fluctuations in the alignment of normals to the surface. The two-point function
decays exponentially
\beq
 <{\bf \hat n}(\xi_1,\xi_2)\cdot{\bf \hat n}(0)> =
e^{ - \frac{ |\xi|}{ \xi_p} }
\eeq
with persistence length $\xi_p$. Thus the surface is always disordered
or {\it crumpled} at length scales $r$ exceeding $\xi_p$. This
conclusion is of considerable interest in the study of liquid
membranes as well. A typical example of a liquid membrane found in
nature (which can also be manufactured in the laboratory) is a lipid
bilayer. It consists of two layers of amphiphilic molecules with polar
hydrophilic heads and long hydrophobic hydrocarbon tails. These
bilayers arrange themselves in thin extended sheets. Within the
bilayer individual molecules are quite free to diffuse, so that the
in-plane elastic constants turn out to be very low. Another candidate
liquid membrane is a monolayer of surfactant molecules at an oil-water
interface in a microemulsion. In fact any flexible interface between
three-dimensional phases is a candidate system for a liquid membrane
model. We see here a beautiful interplay between string theory,
quantum gravity and the statistical mechanics of fluctuating liquid
membranes (figure: other kinds of membranes)

\subsection{The Model of Random Surfaces We used}
We studied the theory defined by the action
\beq
 \protect\label{discform}
	 S = S_{Gauss} + \lambda S_E =
  \sum_{i,j,\mu}C_{ij}(X^{\mu}_i - X^{\mu}_j)^2 +
  \lambda\sum_{\hat{k},\hat
{l},\mu}C^{\hat{k}\hat{l}}(1 -
  n^{\mu}_{\hat{k}}\cdot n^{\mu}_{\hat{l}})\ .
\eeq
Thus, for $\lambda > 0$, we have introduced a ferromagnetic
interaction in the surface normals.  The model defined by this action
has been studied in \cite{CATTER,BAJOWI,BCJW,AMBONE,AMBTWO,CAKORE} and
references therein.

>From (\ref{kijdef}) and the definition of the induced metric, it
follows that equation (\ref{discform}) is a discretization of the
continuum action
\beq
  \protect\label{ourcontact}
  S = \int\sqrt{|{\det {g}}|}(g^{}\partial_i X^{\mu}
  \partial_j X^{\mu}  + \frac{\lambda}{2}g^{ij}h^{kl}K_{ik}K_{jl})
\eeq
Note that the second term in this action is manifestly positive, Weyl
and reparametrization invariant, and that $\lambda$ is a dimensionless
coupling.  So, naively, it is not clear whether it is relevant or not.
If it were relevant, one would then anticipate that (since it obeys
all of the appropriate symmetries) it should be effectively generated
in any string action, and that it should engender ordering of the
normals. It should then lead to another RG fixed point at a finite
value of $\lambda$, which would characterize a phase transition
between the crumpled phase (observed when $\lambda=0$) and a
renormalization`smooth(er)' phase.\footnote{The extrinsic curvature
term is also higher-derivative, indicating that the field theory
described by this action is non-unitary. This fact alone does not
imply that the associated string-scattering amplitudes do not satisfy
unitarity.  Polchinski and Yang \cite{POLYAN} do, however, contend
that in this case the string theory will not be unitary.  Even if this
were so, this model could still be an appropriate description of the
statistical mechanics of fluctuating surfaces, although not one
corresponding to a physical fundamental string theory.  Braaten and
Zachos \cite{BRAZAC} have also showed that the generic static
classical solutions of a similar higher-derivative theory of rigid
strings are unstable.  This would also imply that these actions could
only be the basis for effective, but not fundamental, quantum theories
of strings.  } We proceed first to review previous work which has
addressed the question of whether or not these theories exhibit a
crumpling transition.

\subsubsection{Previous Analytical Work}
A group analysis \cite{HELFRI,KLEINB,FORSTE,POLEXT} indicates,
however, that there should be no phase transition at finite coupling
when such extrinsic curvature dependent operators are added to the
action.  The computations of refs. \cite{HELFRI,KLEINB,FORSTE} use the
action
\beq
  \protect\label{acthkf}
     S = \int d^2\sigma\ (\mu_0\sqrt{\det{h}} + \frac{1}{\alpha}
    \sqrt{\det{h}}(h^{ij}K_{ij})^2)\ ,
\eeq
in the regime in which the string tension $\mu_0$ is small (unlike the
usual particle physics limit of string theory, which is characterized
by large $\mu_0$). After integrating out fluctuations of the embedding
$X^{\mu}$ between momentum scales $\Lambda$ and $\tilde{\Lambda}$, it
is found that the renormalization of the extrinsic curvature coupling
is given to one-loop order by
\beq
  \beta ({\alpha}) \equiv \Lambda \frac{d\alpha}{d\Lambda}
  = - \frac{3}{4\pi}\alpha ^2 \ ,
\eeq
so that $\alpha$ is driven to infinity in the infra-red.  This theory
thus exhibits asymptotic freedom. Surfaces are smooth (the normals are
correlated) below a persistence length\cite{PELLEI}.
\beq
  \protect\label{persist}
  \xi_p \sim \exp(\frac{4\pi}{3\alpha_{bare}})\ ,
\eeq
and are disordered above this scale.  
% According to these arguments, the
%extrinsic curvature suppressing term fails to eliminate the pathological
%surfaces that make the continuum limit of these theories trivial. Some
Some intuition into this result can be gained by observing that this
theory is similar to the $O(3)$ sigma model, which is asymptotically
free \cite{POLBOO}. The normals to $M$ are the analogues of $O(3)$
vectors, though in this case they are constrained to be normal to a
surface governed by the action (\ref{ourcontact}).

Without the extrinsic curvature term, (\ref{acthkf}) is the Nambu-Goto
action, while (\ref{ourcontact}), which we use in our simulations, is
based on the action quantized by Polyakov.  Classically (when the
equations of motion for the Polyakov action are solved and substituted
back into the action), the two actions are equivalent.  It has also
been demonstrated \cite{MORRIS} that the two quantizations are
equivalent in the critical dimension $D=26$.  In lower dimensions
(note that the Nambu-Goto action clearly does not make sense for $D <
2$), it is not so clear that quantizations `based' on the two actions
are indeed the same.  The work of Polchinski and Strominger
\cite{POLSTRO} suggests that there are alternative quantizations.
Distler (\cite{DIST3D}) has also questioned the equivalence of these
quantizations in $D=3$.  Indeed, even if the two quantizations are
equivalent, it does not automatically follow that the two theories are
still the same once an extrinsic curvature dependent term has been
added.

In fact, Polyakov in \cite{POLEXT} uses a hybrid form of the action
(\ref{acthkf}) and still obtains the same result for the beta
function. He introduces an intrinsic metric $g_{ij}$, chooses the
conformal gauge $g_{ij}=\rho \delta_{ij}$ and considers
\beq
  \protect\label{polyact}
  S = \frac{1}{2\alpha}\int d^2\sigma( \mu_o \rho + \rho^{-1}
  (\partial^2X^{\mu})(\partial^2X^{\mu}) + \lambda^{ij}(\partial_iX^{\mu}
  \partial_jX^{\mu} - \rho\delta_{ij}))\ .
\eeq
Classically, the Lagrange multiplier $\lambda^{ij}$ constrains the intrinsic
metric to equal the induced metric (this equality is not enforced by the
classical equations of motion for the original Polyakov action).  This
constraint should be relaxed quantum mechanically if, as Polyakov
\cite{POLBOO} argues,  the condensate of this Lagrange multiplier assumes a
value of the order of the momentum cutoff.  If this dynamical
assumption is correct, then one can essentially derive the equivalence
of this Nambu-Goto like and the original Polyakov quantizations.  In
the large $D$ (embedding dimension) limit, saddle point calculations
\cite{DAGUAE} show that $\lambda$ indeed does acquire a large
expectation value, and that for small values of the string tension
$\mu_o$, the coupling $\alpha$ is asymptotically free, as the RG
calculations suggest.

There are, however, a couple of caveats and suggestions in the
analytic literature that do allow for the existence of a crumpling
transition for fluid surfaces.  Polyakov remarks that if, in the
infrared region, fluctuations of the internal geometry ($\rho$) are
suppressed relative to fluctuations of the extrinsic metric, then the
beta function is proportional to $\alpha$ and hence the continuum
limit of the theory exhibits non-trivial scaling behaviour; this
presumably cannot be the case in the large $D$ limit.  Another RG
calculation, performed by Yang \cite{YANG} using the
Polchinski-Strominger action \cite{POLSTRO} with an extrinsic
curvature dependent term, indicates that the two-loop correction
(which is proportional to $\alpha^3$) might be large enough to yield a
zero of the beta function, and thus a non-trivial IR fixed point.  The
Polchinski-Strominger action is based on the assumption that the
Liouville mode $\rho$ effectively decouples (its mass is much greater
than the momentum scale set by the string tension); it is not clear
why this assumption should hold for the model that we simulate.
Finally, note that these computations are perturbative (in $1/D$ or
$\alpha$). It is possible that non-perturbative effects could drive a
crumpling transition.

\subsubsection{Previous Numerical Evidence}

Monte Carlo simulations of the action (\ref{ourcontact}) on
dynamically triangulated random surfaces (DTRS) were first performed
by Catterall
\cite{CATTER}, and  shortly thereafter by Baillie, Johnston, and Williams
\cite{BAJOWI,BCJW} and Catterall, Kogut and Renken \cite{CAKORE}.  They
simulated triangulations with the topology of the sphere, and measured the
specific heat

\beq
  C(\lambda) \equiv \frac{\lambda^2}{N}(<S_E^2> - <S_E>^2)\ ,
\eeq
on surfaces with up to $N = 144$ nodes (and $N = 288$ nodes in the
last reference).  They found a peak in the specific heat; the peak
size appeared to grow with $N$.  A similar model that can be
vectorized rather straightforwardly was also considered; the set of
planar $\phi^3$ graphs was simulated \cite{CAKORE,CEKR}.  Each vertex
of these $\phi^3$ graphs was embedded in $R^3$ and the action
(\ref{ourcontact}) was used; graphs of up to $1000$ nodes were
simulated (these would be dual to $500$ node triangulations).  It was
found that the specific heat peak grew with $N$, albeit slowly, as
\beq
\label{specfit}
	    C_{max} = AN^{\omega} + B\ ,
\eeq
with $\omega = 0.185(50)$.  Further work by Ambj{\o}rn, Irb\"ack,
Jurkiewicz, Petersson and Varsted \cite{AMBONE,AMBTWO}, using
dynamical with the topology of the torus and lattices with up to
$N=576$ nodes, indicated that the rate of increase of the peak height
severely diminishes with increasing $N$.  The data strongly suggests
that in fact the specific heat peak height does not diverge as
$N\rightarrow\infty$.  These authors also measured the bare string
tension and mass gap, by embedding the torus in a background toroidal
space spanned by a loop, and measuring the dependence of the free
energy on the loop size.  They found that these measurements (when
taken for $\lambda$ values near the peak position) are consistent with
the appropriate scaling relations (with vanishing bare string tension
and mass gap, i.e. correlation length tends to be infinite) that
should characterize a phase transition to smooth surfaces.  This
measurement, although it constitutes the best evidence there is so far
for a real phase transition at $\lambda=\lambda_c$, is still quite an
indirect way of measuring correlation functions.  As we will discuss,
these scaling relations could contradict other observed phenomena such
as the absence of diverging correlation times and increasing finite
size effects at the putative critical point.

Thus it appears that numerical evidence could allow for the existence of a
crumpling transition (most probably of higher order), while analytical
calculations generally indicate that no such transition should occur.

In \cite{KROGOM} the peak was measured in a DTRS simulation that
incorporated self-avoidance and the extrinsic curvature term $S_E$,
with a solid-wall potential substituted for the Gaussian term in the
action. The results for the specific heat turned out to be very
similar to those found in the simulations we have just discussed, for
example, in \cite{AMBTWO}.  The specific heat peak is, in this
context, considered to be a lattice artifact, because the peak height
levels off with large $N$ (of order $500$).  These simulations
included a crude block-spin measurement that suggests that the
renormalization group flow of $\lambda$ is consistent with the
analytical result of asymptotic freedom.

Simulations using other discretizations for the extrinsic curvature
dependent term have yielded somewhat different results
\cite{CATTER,BAJOWI}.  The specific heat peak, measured in simulations
observablesemploying what is referred to as the `area discretization',
is rather feeble, and levels off for small values of $N$ (by $N=72$)
(the authors interpret this as being indicative of perhaps a `third'
order transition). Actions based on these various discretizations have
been simulated for fixed, triangular meshes.  These systems model
tethered or crystalline membranes, in which the constituent molecules
are tightly bound together.  In the tethered case, the specific heat
peak obtained from simulations of the edge action (\ref{ourcontact})
grows vigorously as a function of $N$ for very large ($128 \times
128$) lattices \cite{WHEATE}.  This is strong evidence for the
existence of a second order transition which, in this case, is in
accord with the analytic results -- these calculations are reviewed by
Nelson
\cite{NELSON} and David \cite{DAVREV,DAVREZ} and involve mean field
and large $D$ computations which suggest that the $\beta$ function is
linear at leading order, with a zero for finite $\alpha$, i.e. a UV
fixed point.  When the alternate area discretization is used in the
tethered case, the specific heat peak again stops growing.  Recent
work has demonstrated that this other discretization is pathological
in the tethered case; the class of `corrugated' surfaces, which are
singular in one direction and smooth in the other, then dominates the
path integral \cite{WHEATE}.

Thus, given the muddle of somewhat contradictory evidence, it is
unclear whether or not a crumpling transition exists for fluid
surfaces.  We have pursued this question by taking high statistics
measurements of the specific heat peak, and by measuring other
 describing the geometry of these surfaces, since
observables with different quantum numbers can give quite different
information.  For example, in the Ising model the magnetization
behaves quite unlike the internal energy (which is invariant under the
standard $Z_2$ transformation).

To analyze and interpret this data, we have applied insights gained
from work on better understood systems, primarily spin models and
lattice gauge theories. Issues of the equivalence of the Nambu-Goto
and Polyakov quantizations have also motivated us to compare the
intrinsic and induced geometry of the surfaces that we simulate.

\subsection{The Goal -- If It Is A Phase Transition Or Other Cases} 

In the last few years random surfaces have been extensively explored
via numerical simulations on a wide range of computers, including
parallel machines \cite{CATTER,BAJOWI,BCJW,AMBONE}.  There are some
novel but not fully understood results. The full action which is
simulated is given by a quadratic interaction term plus the extrinsic
curvature term in equation (\ref{discform}).  For
$\kappa<\kappa_c\simeq 1.5$ one sees the expected crumpled surface
(see fig. $7$). The radius of gyration of these surfaces grows only
logarithmically with their area corresponding to infinite Haussdorf
dimension $d_H$ defined by
\beq
	R_G^2 \simeq A^{{2 \over d_H}}
\eeq
where $R_G$ is the radius of gyration.  For $\kappa>\kappa_c$ the
surfaces become extended and considerably smoother with $d_H$
approaching two, which would be the value one would get for a flat
surface (see fig. $8$).  The nature of the cross-over at $\kappa_c$ is
still uncertain. It may be that the system is undergoing a true
thermodynamic phase transition. If it is of second order then the
continuum limit constructed at the critical coupling would be an
interesting string theory corresponding to a real extended $2d$
surface rather than a branched polymer with its largely
one-dimensional character. In this case it must be that the coupling
${1 \over \kappa}$ ceases to vary with scale (there is a fixed point
of the beta function $q{{d\kappa \over dq}}$) at the critical coupling
$\kappa_c$. At this point there is said to be a {\it crumpling
transition}. This is the most exciting possibility from the string
point of view because it would mean that we have successfully
regularized and defined the quantum theory of the string with more
than one embedding dimension without any instability arising. The
challenge would then be to understand the exact nature of the
continuum string theory at the crumpling transition and the origin of
the fixed point.

It may also be that the observed cross-over is not a true phase
transition and that the persistence length is simply reaching the
finite size of the surface that is simulated on the computer. In this
case it could still be that the surface is always crumpled on
sufficiently large distance scales. This is a real possibility for a
liquid membrane but would still leave us without a viable lattice
regularization of a string in $d>1$ dimensions. We are presently
performing large-scale simulations in three and four embedding
dimensions to decide which of the above possibilities is in fact
correct \cite{BCHHM}.

Finally it is of great interest to extend the technique of dynamically
triangulated surfaces to manifolds of higher dimension, in particular
to three and four dimensional manifolds. One can then simulate say
four dimensional Einstein-Hilbert quantum gravity and seek critical
points which provide a non-perturbative definition of a perturbatively
non-renormalizable quantum field theory. This would be a very exciting
development. Preliminary work indeed seems to indicate that there are
indeed phase transitions in $4d$ gravity \cite{AAABJ}

In my study, the main focus is to numerically study the cross-over
region and try to see if it is true second order phase transition. If
it is corresponding to a critical point, then there is the exciting
possibility of obtaining a well defined continuum string theory for
$d>1$.


Evidence from recent and ongoing numerical
simulations of dynamically triangulated random surfaces indicate that
there is a non-trivial crossover from a crumpled to an extended
surface as the bending rigidity is increased.  


%\include{sec3}
% $Id: sec3.tex,v 1.6 1994/06/25 13:44:06 han Exp $

\newpage

\section{Measurement, Data Analysis And Results}

In Section \ref{S_OBS} we define the quantities we have decided to
measure, and explain why they are physically interesting. Next, in
Section \ref{S_NUM}, we present the details of our numerical
simulations, including a complete high-statistics analysis of the
behavior of a set of relevant observables. Since computing correlation
functions on dynamically triangulated surfaces is a difficult task, we
have focused on elucidating the phase diagram by analyzing local
observables in great detail.

\subsection{Observables \protect\label{S_OBS}}

To minimize finite size effects, we have considered triangulations
with the topology of the torus.  The action (\ref{ourcontact}) was
used, with the BRST invariant measure utilized also by Baillie,
Johnston, and Williams (\cite{BAJOWI}), so that

\beq
  Z = \sum_{G \in T(1)}
  \int\prod_{\mu,i}dX^{\mu}_{i}\prod_{i}q_i^{\frac{d}{2}}
  \exp( -S_{Gauss} - \lambda S_E)\ ,
\eeq
where $d=3$, $q_i$ is the connectivity of the $i$th vertex, and $T(1)$ refers
comto the set of  of genus $1$. The authors of
\cite{AMBONE,AMBTWO} do not include this connectivity dependent term in
their measure.  The long-distance physics of the simulations is
presumably insensitive to the presence of this term.  Because we have
chosen a different measure, though, our quantitative results cannot be
precisely compared with theirs.

We measured a variety of quantities that characterize the extrinsic
and intrinsic geometry of these surfaces.  These observables include:

\begin{enumerate}

\item

  The edge curvature $S_E$ and the associated specific heat
$C(\lambda)$, which is a sensitive indicator of the presence of a
phase transition.
\beq
	C(\lambda) = \frac{\lambda^2} {N} (<S_E^2> - <S_E>^2)\ .
\eeq
This exhibits a peak at a coupling $\lambda_c$ which depends on the
exact discrete form of the action chosen
\cite{CATTER,BAJOWI,BCJW,CAKORE,CEKR,AMBONE,AMBTWO,BCHHM}.

\item

  The squared radius of gyration $R_G$;

  \beq 	
	R_G \equiv \frac{1}{N}\sum_{i,\mu}(X_i^{\mu} - 
        X_{{\rm{com}}}^{\mu})^2\ , 
        \protect\label{E_RG} 
  \eeq 
unthe  subscript refers to the center of mass of the surface.  By
measuring the $N$ dependence of the gyration radius, we can extract a
value for the extrinsic Hausdorff dimension, which is given by

  \beq
    R_G \sim N^{\nu}\sim N^{\frac{2}{d_{\rm{extr}}}}\ \ .
    \protect\label{E_RGD}
  \eeq

\item

  The magnitude of the extrinsic Gaussian curvature. We measure
  a discretization of $\int \mid K \mid \sqrt{\mid h \mid}$,
  with

  \beq
    \mid {\cal{K}} \mid = \frac{1}{N}\sum_i\mid 2\pi -
    \sum_{\hat{j}} \phi_i^{\hat{j}}\mid\ .
    \protect\label{E_CK}
  \eeq
  Here $\phi_i^{\hat{j}}$ denotes the angle subtended by the $\hat{j}$th
  triangle at the $i$th vertex.  This quantity, therefore, measures the
  magnitude of the deficit angle in the embedding space averaged over
  all vertices. We also record the mean square fluctuation of
  $\mid{\cal{K}}\mid$, denoted by $F[\mid{\cal{K}}\mid]$.

\item

  The corresponding intrinsic quantity, $\mid {\cal{R}} \mid$, given by

  \beq
    \mid {\cal{R}} \mid = \frac{\pi}{3N}\sum_i\mid 6 - q_i \mid\ ,
    \protect\label{E_CR}
  \eeq
  and its fluctuations.  When the intrinsic and extrinsic metrics are equal,
  the intrinsic and extrinsic deficit angles are identical, and
  $K = R/2$.

\item

  To study the correlation between intrinsic and extrinsic geometry,
  we also measure the quantity which we refer to as ${\cal{K}}*{\cal{R}}$:

  \beq
    {\cal{K}}*{\cal{R}} \equiv \frac{\int KR}{\sqrt{\int K^2 \int R^2}} =
    \frac{\sum_i(2\pi - \sum_{\hat{j}}\phi_i^{\hat{j}})(6 - q_i)}{\sqrt{
    \sum_i(2\pi - \sum_{\hat{j}}\phi_i^{\hat{j}})^2\sum_i(6 - q_i)^2}}\ .
    \protect\label{E_KR}
  \eeq

  This quantity is $1$ when the metrics are equal, $0$ if they are -
  correlated, and negative when these curvatures are anti-correlated.

\item

  We measure, finally, the average maximum coordination number of the
  surface vertices,  ${\rm{max}}_i \ q_i$.

\end{enumerate}

\subsection{The Numerical Simulation \protect\label{S_NUM}}

In this chapter, I will mainly describe the physics part implemented
in the computer program and leave the explaination of the computer
program in capter 5 with other computational issues together.

In our simulations we have used the standard Metropolis algorithm to
update the embedding fields $X^\mu_i$.  To sweep through the space of
triangulations we performed flips (see reference \cite{EALRYM}) on
randomly chosen links.  Flips were automatically rejected if they
yielded a degenerate triangulation; i.e. one in which a particular
vertex has fewer than three neighbors, or in which a vertex is labeled
as its own neighbor, or where more than one link connects two
vertices. (It has been proven in ref.
\cite{EALRYM,MINOEX} that the entire space of graphs of a given topology can
be spanned by only performing these flips.)  After a set of  $3N$ flips was
performed, $3N$ randomly selected embedding coordinates were updated via
random shifts from a flat distribution,

\beq
  X^{\mu} \rightarrow X^{\mu} + \delta X^{\mu}\ .
\eeq

The mean magnitude of these shifts

\beq
  <\delta X^{\mu} \delta X^{\mu}>
\eeq
was chosen so that the acceptance rate for updates of the $X^{\mu}$
was roughly $50$ percent.  Most of the Monte Carlo simulations were
performed on HP-9000 (720 and 750 series) workstations; we also
collected some data by binnedsimulating lattices on each of the 32
nodes of a CM-5.  Our code was in Fortran, with a Fibonacci random
number generator.

\begin{table}
\begin{tabular}{|l||l|l|l|l|l|l|l|l|} \hline N=36 &&& 1.375 & 1.400
& 1.425 & 1.475 && $\lambda$\\ \cline{2-8} &&& 3 & 3 & 3 & 3 &&
$\times 10^6$ sweeps \\ \hline N = 72 &&& 1.375 & 1.400 & 1.425 &
1.475 && $\lambda$\\ \cline{2-8} &&& 3 & 3 & 3 & 3 && $\times 10^6$
sweeps \\  \hline
N=144 &.8 &1.25 & 1.35 & 1.40 & 1.45 & 1.50 & 2.0 & $\lambda$\\ \cline{2-8}
 & 3 & 3 & 3 & 3 & 3 & 3 & 3 & $\times 10^6$ sweeps \\ \hline
N=288 & .8 & & 1.375 & 1.40 & 1.425 & 1.475 & 2.0 & $\lambda$\\ \cline{2-8}
 & 14.4 && 21.0 & 15.0 & 16.2 & 13.5 & 14.4 & 
$\times 10^6$ sweeps \\ \hline
N= 576 & .8 & 1.325 & 1.375 & 1.40 & 1.425 & 1.475 & 2.0 & $\lambda$
\\ \cline{2-8}
& 12.0 & 27.0 & 27.0 & 27.0 & 27.0 & 27.0 & 9.6 & $\times 10^6$ sweeps \\ \hline
N= 1152 & &  &  &  & 1.425 & 1.430 & 1.435 & $\lambda$
\\ \cline{2-8}
&  &  &  &  & 54 & 21 & 18 & $\times 10^6$ sweeps \\ \hline
N= 2304 & &  & 1.40 & 1.42 & 1.425 & 1.430 & & $\lambda$
\\ \cline{2-8}
&  &  & 5 & 5 & 17 & 5 &  & $\times 10^6$ sweeps \\ \hline
\end{tabular}\protect\caption[CT_ONE]{A record of the number of sweeps performed
at each different $\lambda$ value for different lattice sizes.
\protect\label{T_ONE}}
\end{table}

In Table (\ref{T_ONE}) we summarize our runs. All these runs were done
in two major stages. At the stage one, with the speed of our initial
Fortran program, the feasible lattice size we could run was up to
N576. After the runs, we found out that the results were not
conclusive. Since the result was not at all clear whether a true
continuous thermodynamic phase transition separated the two regimes.
There could be a several alternative interpretations.  Perhaps the
simplest possibility, advocated in
\cite{KROGOM}, is that the persistence length $\xi$ describing the
exponential decay of the normal-normal two-point function in the
crumpled (disordered) regime simply reaches the finite size of the
system at $\lambda_c$.  In this case the observed smooth regime would
be a finite-size artifact with the true continuum theory really being
crumpled for all couplings $\lambda$, in accordance with perturbative
analytical results
\cite{HELFRI,KLEINB,FORSTE,POLEXT}.  Since $\xi$ grows exponentially
with $\lambda$, according to the one-loop beta-function, this
interpretation would imply that $\lambda_c$ diverges logarithmically
with system size $N$.  To resolve this issue and to gain further
insight into the model it was clearly desirable to extend the
numerical simulations to larger lattice sizes and to clarify the
influence of finite-size effects. 

Therefore, we optimized our program. The new program has much faster
speed. Within reasonable period of time, we ran up to lattice size up
to N2304. Note that we have performed quite long runs on the larger
lattice sizes. We will discuss in Section (\ref{S_TIM}) why we believe
runs of this length are just sufficient to yield accurate values of
the observables for the largest lattice size ($N=2304$). (as of
lattice meeting 1993 see review about RS).

In all of our figures the different dots will be printed with their
associated statistical error (sometimes too small to be visible). The
statistical error is computed by means of a standard binning
procedure. We will explicitly discuss the cases in which our estimator
for the statistical error is not asymptotic.

The lines in these figures are from a histogram reconstruction (see for example
\cite{ENZOHI,SWEFEA}). We patch different histograms
\cite{SWEFEB,ALBESA,SWESTA} by weighting them with the associated statistical
reconstructions in determination (which we estimate by a jack-knife
procedure); this procedure seems to be very effective and reliable.
All of the reconstruction curve sets ($3$: dotted, dashed and
continuous for $3$ surface sizes on each figure) consist of $3$ curves
(which sometimes appear as a single one). The middle curve is the
histogram reconstruction, and the upper and the lower ones bound the
data within the errors obtained by the procedure we have just
described.

In our new series of simulations on lattices, for $N=144$ we have
patched the four histograms originating from $\lambda=1.35$, $1.40$,
$1.45$, $1.50$. For $N=288$ we have used $\lambda=1.375$, $1.40$,
$1.425$ and $1.475$. For $N=576$, we chose $\lambda=1.375$, $1.40$ and
$1.425$. With $1152$ nodes we ran $54$ million sweeps at
$\lambda=1.425$, $21$ million sweeps at $\lambda=1.430$ and $18$
million sweeps at $\lambda=1.435$
\footnote{ Since the autocorrelation time $\tau$ is of order $400,000$
sweeps on the $1152$ lattice these runs have at least $45 \tau$
measurements.}.  On the data from these three points we use
multi-histogram reconstruction
\cite{ENZOHI,SWESTA}. This works well in that three different reconstructions
give coherent results.  On lattices of $2304$ nodes we have poorer
statistics. We ran $17$ million sweeps at $\lambda=1.425$ plus
approximately $5$ million sweeps at $\lambda = 1.40$, $1.42$ and
$1.43$ as a consistency check.  On the $2304$ lattice histogramming
does not work well.  This is to be expected since the statistics are
not good enough for such a large lattice. Still we have checked that
our measurements at $\lambda=1.425$ give consistent results, that the
error estimate is reliable and that we are, with good accuracy, at the
peak of the specific heat.  In all these simulations for $1152$ and
$2304$ required the equivalent of approximately one year of CPU time
on an HP 9000 (720 series) workstation. (for N1152 and N2304, we also
measured all observables we described above, but we only ploted those
of most physical interest.

Since there are possibilities that could account for the observed
behaviour of $C(\lambda)$ without invoking a phase transition.  One
was based on the analogy between the present model and the $O(3)$
sigma-model in $2d$ \cite{ZINJUS}.  This model is also asymptotically
free and consequently disordered at all non-zero temperatures.  Yet
numerical simulations show a distinct peak in the specific heat which
grows for small lattices and then saturates, just as we find in the
model of a rigid string treated here. This may be seen in Fig.~2 where
we have plotted the specific heat $C(\beta)$ for the two dimensional
$O(3)$ model. The simulations were done on square lattices of volume
$N=16$, $25$, $64$, $100$, $900$, $2,500$, $4,900$ and $10,000$ using
the Wolff algorithm
\cite{WOLFFO}. For each point of the $N=25,100$ and $10,000$ lattices
we used $100,000$ measurements. We took a measurement every time the
Wolff clusters updated a volume exceeding $30$ times the volume of the
lattice \footnote{For the $N=16$, $64$, $900$ and $4,900$ lattices the
integrated autocorrelation times were between $1$ and $2$ Wolff
updatings of the entire lattice}.  For each point of the $N=16$, $64$,
$900$, $2,500$ and $4,900$ lattices we used $20,000$ measurements.  We
took a measurement every time the Wolff clusters updated a volume
exceeding $3$ times the volume of the lattice.  It is very clear that
the peak levels off quickly for $N\ge 100$ and that ``$\beta_c$'' is
not increasing with the size of the lattice.  Measurements of the
asymptotic value of $C(\beta)$ have been reported in the past
\cite{MAPAPE,Colot}. The authors of
\cite{BRDE,BRDESI,ORBR} explain the peak as the excitation of an
extra degree of freedom, the so-called $\sigma$-particle
\cite{BALESH}. The would-be  transition occurs when the mass of the
$\sigma$-particle becomes comparable to the inverse correlation length
of the $O(3)$ model. It may be that there is a similar
interpretation of the observed peak of $C(\lambda)$.

We have only drawn the reconstructed, patched curves (with their
reliable errors) in the regions where we trust them. For example,
close to the pseudo-critical region we can trust a peak pattern only
when we can reconstruct the peak by starting from both sides of the
transition (without multi-histogram patching). So we have always used
single histogram to check these criteria, before constructing the
final, multi-histogram data.

\subsection{The Phase Diagram \protect\label{S_PHA}}

We have measured, as stated previously, a large number of local
observables. We will see that a mixed picture emerges from these
measurements. For example the observables related to the dynamical
triangulations exhibit a characteristic pattern, to be discussed in
detail below.

We start by showing, in Fig.1, the edge curvature $S_E$ as a function
of $\lambda$.  The crossover region is around $\lambda \simeq 1.4$.
For small values of $\lambda$, the surface is crumpled (see the latter
part of this section). In this region, finite size effects are already
negligible for our lattice sizes, and our $3$ data points are on top
of each other. We can see weak finite size effects by comparing the
continuous lines in the transient region. The $N=144$ dotted line is
far from the ones of the two larger lattices, which lie, on the
contrary, on top of each other. Finite size effects are larger in the
large $\lambda$ phase. One would expect, close to a phase transition
with a diverging correlation length, an increase in finite size
effects which we do not observe here. The lattice should feel the
presence of the zero mass excitation, and the finite size corrections
should be larger than everywhere else (in the case of periodic
boundary conditions they would obey a power-law, rather than decaying
exponentially with size). This is surely not firm evidence against the
presence of a phase transition, but it does show that the
putative critical behaviour is atypical.

The errors in the `flat phase' ($\lambda=2.0$) are not under control.
Our estimators do not plateau under repeated iterations of the binning
procedure.  In this regime, correlation times are large, as we will
discuss in next section. This {\em caveat} holds for this figure and
for all the quantities we have measured.

The specific heat peak for $N=576$, $1152$ and $2304$ is shown in
Fig. 1a.  In Table \ref{T_TWO} we give our results for the maximum of the
specific heat and the associated coupling $\lambda_c$ as a function of
$N$ \footnote{We have reanalyzed the data presented in reference
(\cite{BCHHM}), using a different method of weighting relative errors
when combining histograms.  Thus, some of the errors quoted here are
smaller than the respective uncertainties in reference
(\cite{BCHHM}).}.

In Fig. 2 we show the related specific heat $C(\lambda)$, in the same
$\lambda$ region. In Fig. 3 we enlarge the pseudo-critical $\lambda$ region,
in order to show the reconstructed peak of the specific heat. As already
noted our reconstruction procedure is quite reliable here.

In Table \ref{T_TWO} we give the maximum of the edge curvature specific heat
and its location for different lattice sizes.

\begin{table}
\begin{center}
\begin{tabular}{||r||r|r||} \hline
$N$ & $C^{(max)}$ & $\lambda_c$ \\ \hline
 $  36$ & $3.484 \pm .008$ & $ 1.425 \pm .035$ \\ \hline
 $  72$ & $4.571 \pm .015$ & $ 1.410 \pm .015$ \\ \hline
 $ 144$ & $5.37  \pm .08 $ & $ 1.395 \pm .017$ \\ \hline
 $ 288$ & $5.55  \pm .05 $ & $ 1.410 \pm .015$ \\ \hline
 $ 576$ & $5.81  \pm .06 $ & $ 1.425 \pm .010$ \\ \hline
 $1152$ & $5.69  \pm .04 $ & $ 1.425 \pm .010$ \\ \hline
 $2304$ & $5.75  \pm .10 $ & $ 1.425 \pm .010$ \\ \hline
\hline
\end{tabular}
\end{center}
\protect\caption[CT_ONE]{The maximum of the specific heat and its position,
with errors, for different lattice sizes.
\protect\label{T_TWO}}
\end{table}

We see that the specific heat peak grows vigorously with $N$ for small
lattices, appears that there was a crumpling transition very similar
to that of the crystalline surface. However on larger lattices the
specific heat height growth levels off for larger $N$.  From the data
for the three largest lattices, we can extract a specific heat
exponent $\omega = .06\pm .05$, with $\omega$ defined as in equation
(\ref{specfit}), and the constant $B$ set to zero.  If we estimate an
effective exponent from the $N=144$ and $288$ lattices, we get
$.05\pm.06$, and from the two largest lattice sizes we get
$.07\pm.06$; this demonstrates that we do not see, within our
statistical precision, any sign of a non-pure-power, non-asymptotic
behavior. Note that, if the constraint that $B$ vanishes is relaxed,
our data is not accurate enough to yield a meaningful fit to equation
(\ref{specfit}).  A very small (asymptotically finite) correlation
length is sufficient to produce such a small effect on our quite small
lattice sizes. These results appear to be consistent with those of the
Copenhagen group \cite{AMBTWO}, and they are not so far from the ones
of the Urbana group \cite{CAKORE,CEKR}.

The critical value of $\lambda$ shifts very slowly to higher values
for increasing $N$ for the largest lattices, although the increase is not
statistically significant.

In addition the shape of the specific heat (for example the width) is
basically unchanged as we go to larger lattices. From Figs. 2 and 3 we
do not infer evidence of criticality.

In Fig. 4a we show the radius of gyration of the surface, $R_G$, as defined
in (\ref{E_RG}). Here obviously the volume scaling is non-trivial: larger
surfaces have larger radius. The histogram reconstruction already ceases
to work for quite low values of $\lambda$ for the larger lattice.
This effect could be related to the interesting finite size scaling behavior
of this quantity, which we illustrate in better detail in Fig. 4b.
Here we plot
\beq
  \nu(N) \equiv \frac{\log \frac{R(N)}{R(\frac{N}{2})}} {\log(2)}\ .
\eeq
This is an effective inverse Hausdorff dimension, which is a function of
$\lambda$. In the large $\lambda$ limit $\nu \to 1$ and $d_{extr}\to 2$, as
expected for flat surfaces. In the low $\lambda$ limit $d_{extr}$ becomes
very large. In the pseudo-critical region $\nu$ is a linear function of
$\lambda$. Curiously enough, the latter curve yields a Hausdorff
dimension of $4$, a value characteristic of branched polymers, near the
location of the specific heat peak.
This value is not particularly reliable though because of finite-size effects
and also because it changes rapidly in this region.  In
ref. \cite{AMBTWO} a value compatible with ours ($D_H(\lambda_c)>3.4$)
is quoted for the critical theory.
We stress however (and also here we are in complete agreement with
\cite{AMBTWO}) that the  dimension in the pseudo-critical
region depends heavily and quite unusually on $N$.

In both the high and low $\lambda$ regions finite size effects are
quite small (compatible with zero to one standard deviation).
In the pseudo-critical region, on the contrary,
finite size effects are large. This effect
cannot be explained by the shift in $\lambda$ which one gets from the shift of
the peak of the specific heat, which is far too small. This behaviour is very
different from that we discussed for $S_E$ and it seems to
indicate the possibility of some sort
of critical behavior close to $\lambda=1.4$.

With our new runs on larger lattice, we note the behaviour of the
gyration radius at $\lambda_c$. For large $\lambda$ ($>2$), the
scaling of $R(N) \sim N^{\frac{2}{d_H}}$ with $N$ gives a Hausdorff
dimension close to $2$ (as we expect for flat surfaces). In the
crumpled region the Hausdorff dimension rapidly increases with
diminishing $\lambda$.  We had pointed out that finite size effects
were relevant in the sector close to $\lambda_c$ and that we could not
estimate a reliable number from the lattice sizes analyzed. Here the
largest lattice we simulated ($N=2304$) does not give useful data,
since the error in $R$ is too large, but on the $1152$ and $576$ node
lattices we get a fairly precise estimate of $R$, which allows us to
estimate for the Hausdorff dimension at the pseudo-critical point
$\lambda_c$ the value $d_H = 4.35 \pm .3$.  This is an intriguing
result, since $4$ is the extrinsic Hausdorff dimension of a class of
branched polymers, as constructed, for instance, in references
{\cite{AMBBP}}.  such configurations are expected to dominate the
string functional integral for large embedding dimension $D$.

In Fig. 5 we plot the expectation values of the magnitude of the extrinsic
Gaussian curvature  $\mid {\cal K} \mid$. If the induced metric is equal
to the intrinsic metric, then $\mid {\cal K} \mid =
\frac{\mid {\cal R} \mid}{2}$.

This plot is not substantially different from that of $S_E$. We note
that finite size effects are somewhat larger in this case than for the
edge action, but they follow the same pattern (exhibiting a big
increase in the flat phase).

The plot of the fluctuations of the extrinsic Gaussian curvature,
$F[{\cal{K}}]$, which we
present in Fig. 6, shows
something very new. A very sharp crossover, with perhaps a
peak developing for large $N$, dominates the pseudo-critical behavior.
Fluctuations do not seem to depend on $\lambda$ in the crumpled phase, while
they drop dramatically, in a very small $\lambda$ interval, in the flat
region. Here again, finite size effects are sizeable in the pseudo-critical
region. The position of the crossover does not depend sensitively
on $N$, while the
detailed shape at $\lambda_c$ seems to change slightly with $N$.

With the new runs on larger lattices, We have also measured the
fluctuations of the extrinsic Gaussian curvature $|{\cal K}|$, and
computed the fluctuations of the mean defect coordination number $|q -
6|$.  We find that on larger lattices the fluctuations of these
observables at $\lambda_c$ also do not grow with $N$; thus their
behaviour does not provide unequivocal evidence of the presence of a
phase transition.

In Table~2 we give the mean-square fluctuations
of both observables.
\begin{table}
\begin{center}
\begin{tabular}{||r||r|r||} \hline
$N$ & $F[K]$ & $F[q-6]$ \\ \hline
 $ 576$ & $5.71  \pm .08 $ & $ 8.39 \pm .04$ \\ \hline
 $1152$ & $5.59  \pm .05 $ & $ 8.32 \pm .03$ \\ \hline
 $2304$ & $5.70  \pm .10 $ & $ 8.37 \pm .06$ \\ \hline
\hline
\end{tabular}
\end{center}
\protect\caption[CT_TWO]{The mean square fluctuations of the extrinsic
Gaussian curvature ${\cal K}$ and the defect coordination number $q-6$,
with errors, for different lattice sizes.
\protect\label{TT_TWO}}
\end{table}

It is difficult to give a precise interpretation of a plot like this, but,
as we said, the crossover is very clear here.

In Fig. 7 we give the intrinsic curvature ${\cal{R}}$ and in Fig. 8 its
fluctuations. Both plots are very similar to the related,
extrinsic curvature,
$\cal{K}$ plots.
$\mid {\cal{R}} \mid$ drops off rapidly, just as $\mid {\cal{K}} \mid$
does.
Through the peak region, though, $\mid {\cal{K}} \mid$ decreases
by about a factor
of $5$ while $\mid {\cal{R}} \mid$ diminishes to only about $.6$ of its value
on the left-hand side of the peak.  Since the action explicitly suppresses
mean curvature, and the mean and extrinsic
Gaussian curvature are closely related
(for instance, $H^2 > \frac{K}{2}$), we would expect that for large
$\lambda$  extrinsic fluctuations would be suppressed much more than
fluctuations of intrinsic geometry.

In Fig. 9 we plot the intrinsic extrinsic curvature correlation. The plot of
${\cal K}*{\cal R}$ indicates that intrinsic and extrinsic geometry are
strongly correlated for small $\lambda$, but as one passes through the peak
observablesregion they become decorrelated.  This
is not particularly surprising, given
that the action directly suppresses only extrinsic fluctuations.
Note that RG calculations based on the Nambu-Goto action plus an
extrinsic curvature term (with no dependence on an intrinsic metric)
perturb about a background that is both intrinsically and extrinsically
flat.  Given the observed decorrelation
between intrinisic and extrinsic geometry,
we would not anticipate that this background appears in the low-
temperature limit of the model which we simulate.

In Fig. 10a we plot the expectation value of the maximum coordination
number, which has non-trivial scaling behavior. In Fig. 10b we give
its scaling exponent, defined analogously to the exponent we have
exhibited for the gyration radius.  In the pseudo-critical region
$q_{max}$ scales (for our $3$ lattice sizes) as a power, with an
exponent close to $0.1$; we do not know if this scaling is meaningful.

\subsection{Correlation Times \protect\label{S_TIM}}

We will discuss here correlation times for different . As we
already pointed out correlation times become very large in the large $\lambda$
region. In agreement with ref. \cite{AMBTWO} (see their Fig. 1) we do not see
any increase of the correlation times close to the pseudo-critical point.

We will not present precise estimates of correlation times
(exponential or integrated) -- they are too large to get precise
estimates. We will limit ourselves to a discussion of a few figures,
which give quite a clear idea of what is happening. The comparison
with Fig. 1 of ref. \cite{AMBTWO} cannot be very direct, since our
action is different, and because their dynamics may be more effective
than ours. Still, the comparison is quite puzzling, since we estimate
and exhibit correlation times which are much (orders of magnitude)
larger than the ones of \cite{AMBTWO}.  Applying customary methods to
estimate $\tau_{int}$ can lead to an underestimate of correlation
times if more than one time scale is present (that does surely happen
with our data if we integrate it on a window of reasonable size).

In Fig. 11a we plot $S_E$ for $N=144$, $\lambda=1.4$, and in Fig. 11b
the gyration radius for these values (with a different time scale).
Clearly, the correlation time is at least of order $40,000$ sweeps in
the first case and $100,000$ sweeps in the second one. In Figs. 12a
and b, we plot the same quantities for $\lambda=1.5$.  Here
correlation times are larger, of order $50,000$ steps for $S_E$ and
larger than $150,000$ steps for $R_G$. In Figs. 13 we draw the same
plot on the largest lattice we study ($N=576$) for $\lambda=1.4$. Here
we can see dramatic correlations, with times of at least $100,000$
steps for $S_E$ and of at least $1,000,000$ steps for $R_G$.

In Fig. 14 we plot, for the same time history and on the same scale,
both $S_E$ and $R_G$. This figure shows a clear anticorrelation:
larger surfaces are flatter and have smaller curvature (this is
apparent in the region close to the 2000th step).

(maybe expand a little further in this chapter)


%\include{sec4}
% $Id: sec4.tex,v 1.4 1994/06/25 13:44:06 han Exp $

\newpage

\section{Conclusions from Our Simulations}
\subsection{Interpretation of Results \protect\label{S_RES}}

On lattices with up to $576$ vertices we can clearly see a sharp
crossover, but the absence of a diverging specific heat, of diverging
correlation times and of strong finite size effects suggests that we are not
observing a usual second order phase transition. On the other hand,
quantities like the intrinsic curvature, or the mass measurements of
\cite{AMBTWO} show that something non-trivial is happening.

Let us review the crux of our observations again.  This model of
crumpled surfaces appears to exhibit sharp crossover behavior in the
region around $\lambda = 1.4$.  The sharp growth in the gyration
radius and the suppression of curvature fluctuations indicate that the
normals acquire long-range correlations, up to the size of the systems
we examine.  Presumably the zero string tension measurement of
\cite{AMBTWO} also shows that the disordered regime differs from the regime in
which the surfaces are ordered (up to scale of the lattices that are
simulated) by only a small shift in $\lambda$. This evidence might
indicate the presence of a phase transition at this point.  If so, it
is very likely to be of order higher than $2$ (or, rather implausibly,
it could be second order with an extremely low negative specific heat
exponent; our lattices are much too small for us to confidently
extrapolate the value of the specific heat exponent as $N \rightarrow
\infty$).

If the transition were higher order, the peak should exhibit a cusp,
but we would need far more accurate data to detect this.  The
existence of this phase transition would then suggest the existence of
a new continuum string theory, though many other issues would have to
be resolved (e.g. unitarity) to determine if such a theory is
physically desirable.

There are other possible interpretations of our data.  We need to
consider the influence of finite-size effects, since the surfaces
which we simulate are quite small, even smaller than one might naively
assume because they are not intrinsically smooth.  For instance,
random surfaces characteristic of $D=0$ gravity have a Hausdorff
dimension of roughly $d_{intr} = 2.8$ \cite{KAWNIM,Migtalk}; it has
been predicted that surfaces embedded in $1$ dimension have Hausdorff
dimension $2 + \sqrt{2}$
\cite{KAWNIM}.  Thus, for instance, if the surfaces in our simulations had an
intrinsic dimension of $3$, they would have a linear size of fewer than $9$
lattice spacings\footnote{Of course, our lattices are too small,  by one or
two orders of magnitude, to really exhibit a convincing fractal structure.}.

Perhaps the simplest alternative explanation for the presence of this
peak is suggested by the arguments of Kroll and Gompper \cite{KROGOM}.
They argue that the peak occurs when the persistence length of the
system approaches the size of the lattice ($\xi_p \sim
N^{\frac{1}{d}}$)\footnote{For couplings above this point, our
simulations would simply be measuring finite size effects.}. For
larger $\lambda$, fluctuations on a larger scale become more
important, but when this scale is greater than the lattice size, these
fluctuations are suppressed. Thus one might surmise that the specific
heat will drop for large $\lambda$. (It clearly goes to zero for small
$\lambda$ because of the presence of the prefactor $\lambda^2$; the
lattice implements a ultraviolet cutoff that freezes out very
short-range fluctuations). However, as shown in section \ref{truep},
we excluded this possibility from our measurements.

The one-loop renormalization group calculation (\ref{persist})
predicts that the persistence length grows as $\xi_p \sim
\exp(C\lambda)$; $C$ is inversely proportional to the leading
coefficient of the beta function.  We would expect that the peak
position should shift to the right with increasing $N$ in this
scenario as

\beq
  \lambda_{peak}(N') - \lambda_{peak}(N) = 
\frac{\ln(\frac{N'}{N})}{d_{intr}C}.
\eeq

Quite a large value of $C$ is needed to explain the rapid crossover;
roughly values of $C \sim 10, d_{intr} \sim 3$ are more or less
consistent with the magnitude of the peak shift and crossover width.
The RG calculations using different forms of the action yield $C =
\frac{4\pi}{3}$ (see equation
\ref{persist}), but
this may not apply to the action we simulate.

This reasoning also indicates that the peak should widen as the
lattice size increases; we do not observe this at all. It seems
plausible though that these arguments, based only on the leading term
of the high lambda expansion, are too naive.

An alternative scenario, which builds on the ideas in the above
paragraph, is suggested by the tantalizing similarities between the
results of our fluid surface simulations and what has been observed
for the $d=4$ $SU(2)$ Lattice Gauge Theory \cite{ENZOHI} and for the
$d=2$ $O(3)$ model. Let us discuss the case of the $O(3)$ model.

The $O(3)$ model, which is asymptotically free, exhibits a specific
heat peak near $\beta = 1.4$.  This peak was first measured via Monte
Carlo simulations by Colot \cite{Colot}. It can also be obtained by
differentiating the energy data measured by Shenker and Tobochnik
\cite{ShTob,Brout}.  The origin of this peak is understood
\cite{Brout,Brout2}; it is due to the fluctuations of the sigma
particle, a low-mass bound state of the massless $O(3)$ pions.  The
sigma induces short-range order, and contributes to the specific heat
as a degree of freedom only at high temperatures (when the correlation
length in the system becomes smaller than its inverse mass).  The peak
thus occurs at the beginning of the crossover regime, when the
correlation length is several lattice spacings.

According to the low temperature expansion, the correlation length
grows as $\xi \sim \exp(2\pi\beta)/\beta$. Thus one would expect a
fairly rapid crossover in the $O(3)$ model; the correlation length
should increase by roughly a factor of $9$ when $\beta$ is shifted by
about $.35$\footnote{In fact, the presence of the sigma significantly
modifies this low-temperature expansion result \cite{Brout2} in this
intermediate regime, but does not qualitatively destroy the rapidity
of the crossover.  Indeed, despite heroic efforts, it has been
impossible to extend computationally beyond this regime and precisely
verify the asymptotic low-temperature relation for the correlation
length \cite{Apost,Hasen}.}.  Such a crossover is indeed observed,
though the histogramming is not so apparent that it is as dramatic as
the crossover behaviour observed for fluid surfaces.\footnote{To
quantitatively compare the width of the crossover regimes for these
two models it would be necessary to measure a correlation length
(perhaps extracted from the normal-normal correlation function) in
these random surface simulations.}

Our results indicate that the specific heat peak grows significantly
when the lattice size $L$ is small, and that virtually no growth in
peak height is evident as $L$ is increased. Also, the peak position
shifts to the right as $L$ grows, and then appears to stabilize for
large $L$.  This is more or less what we observe in our simulations of
fluid surfaces, on lattices of small size.  We point out these
similarities largely to emphasize that there does exist an
asymptotically free theory (with low mass excitations) which exhibits
crossover behavior qualitatively similar to that observed in our
simulations.

The analogy is perhaps deeper, though, since the fluid surface action
(with extrinsic curvature) in certain guises looks like a sigma model
action.  So, perhaps it would not be so surprising from this point of
view to find a sigma particle in these theories perhaps associated
with ($\hat{n}^2 -1$), in which $\hat{n}$ denotes the unit normal to
our surfaces.

Another additional possibility is that fluctuations of the intrinsic
geometry (the Liouville mode) are responsible for short-range order
and contribute to the specific heat peak.

\subsection{True Phase Transition? \protect\label{truep}}

Clearly the maximum of the specific heat curve $C_{max}$ is
effectively constant for surfaces with $576$ or more nodes.  The
(pseudo)-critical coupling $\lambda_c$ is also constant for $N=576$
and above.  With the present data we can definitely exclude the
presence of a divergence in the specific heat. The growth of the
specific heat peak observed on small lattices
\cite{CATTER,BAJOWI,BCJW} does not reflect true asymptotic behaviour.
These results also invalidate the interpretation raised in the
introduction \cite{KROGOM,BCHHM}.  A one-loop renormalization group
calculation shows that the persistence length $\xi$ grows with bending
rigidity $\lambda$ as ${\rm exp}({\frac {3} {4\pi}} \lambda)$.
Equating $\xi$ with the spatial extent of the lattice $N^{1/d_{in}}$,
where $d_{in}$ is the intrinsic Hausdorff dimension of the lattice,
one sees that they become comparable at a coupling $\lambda_c \sim
{\frac {3} {4\pi d_{in}}} {\rm ln} (N)$.  In the continuum limit $N
\rightarrow \infty$, $\lambda_c$ diverges.  Since, for reasonable
values of $d_{in}$, we do not see the increase in $\lambda_c$ with $N$
predicted by the above relationship we can state with some confidence
that the origin of the observed specific heat peak is not explained by
the persistence length becoming comparable to the extent of the
lattice.

It is still possible that there is a true continuous phase transition,
the crumpling transition, occurring at $\lambda_c$. Assuming a
continuous transition, a standard finite-size-scaling argument only
tells us that $\omega= \frac{\alpha} {\nu d} < 0$, where $\alpha$ is
the exponent governing the divergence of the specific heat,
$C(\lambda) \sim {\vert\lambda - \lambda_c\vert}^{-\alpha}$, $\nu$ is
the analogous exponent for the correlation length and $d$ is the
intrinsic Hausdorff dimension of the surface.  In other words there
may be a cusp singularity at $\lambda_c$ as, for example, in the case
of the superfluid ($\lambda$) transition in $He^4$
\cite{LIPA,GOLBOOK}, for which $\alpha = -0.0127 \pm 0.0026$.  Since
we do not have a measurement of $\nu d$, which may even be rather
large, we have no reliable idea of the exponent $\alpha$ itself.
Generally speaking one finds that second order transitions on fixed
lattices become higher order on dynamical lattices, as for example in
the case of the $2d$-Ising model \cite {KAZIS,BOUKAZIS}.  Since there
seems to be a $2$nd order crumpling transition for non-self-avoiding
tethered (fixed-triangulation) surfaces
\cite{KANNEL,AMDUJO,WHEATE,WHESTE},
it would be consistent for the transition to be
higher than $2$nd order when the model is coupled to gravity.

\subsection{Conclusion and Comments}
We have thus explored the phase diagram of fluid random surfaces with
extrinsic curvature. With lattice size up to N576, we were not able to
determine if our model undergoes a phase (crumpling) transition at
finite coupling.  We have observed dramatic crossover behavior for
particular observables in our Monte Carlo simulations, but on the
other hand, the correlation times and certain finite-size effects do
not behave as one would expect in the presence of a phase transition.
With larger lattice up to N2304, we find that the extrinsic-curvature
specific heat peak ceases to grow on lattices with more than $576$
nodes and that the location of the peak $\lambda_c$ also stabilizes.
The evidence for a true crumpling transition is still weak. 
If we assume it exists we can say that the finite-size scaling
exponent $\frac {\alpha} {\nu d}$ is very close to zero or negative.
On the other hand our new data does rule out the observed peak as
being a finite-size artifact of the persistence length becoming
comparable to the extent of the lattice.

The behavior of other lattice models also indicates that it is
possible that we are observing the effects of finite-mass excitations
on small lattices, rather than a phase transition.  We hope that
future work will clarify this murky state of affairs, to determine if
there indeed exists a crumpling transition for fluid surfaces.

All told our work gives only weak evidence for a continuum
crumpling transition. The strongest evidence in favour of
such a transition, at present, is the scaling behaviour of the
string tension and mass gap reported in \cite{AMBTWO}.
This highlights the need for more extensive measurements on these
important observables.

\subsection{Further Work \protect\label{S_WHA}}

There remains much to be done to clarify whether or not a crumpling
transition occurs for a finite value of the extrinsic curvature
coupling $\lambda$.  It would be interesting (and probably a fair
amount of work) to apply Wilson renormalization group techniques to
the actual action (\ref{ourcontact}) which we simulate, to determine
the leading coefficient of the beta function.  Additionally, perhaps a
calculation of $1/D$ corrections to the large $D$ computations already
performed could unearth evidence of a sigma-type excitation in these
theories (the effects of the sigma appear as $1/N$ corrections in the
$O(N)$ model).

We also are our data to examine the behavior of complex zeroes (in
complex $\lambda$ space) of the partition function of our simulations
\cite{winprog}.  It has been shown (in the case of $SU(2)$ lattice
gauge theory) that such zeroes, when they are near but do not approach
the real axis in the infinite volume limit, occur in theories which
exhibit specific heat peaks with no associated phase transition
\cite{ENZOHI}. Low temperature expansions also indicate that the
$O(3)$ model susceptibility has a complex singularity near the real
axis \cite{Butera}-- presumably this corresponds to a zero of the
partition function and is a manifestation of the sigma.

Of course, simulations on large lattices, with better statistics,
should also help us evaluate whether a crumpling transition exists.
We could use parallel computers. However, as I will explain in chapter
5, the real parallelization of random surfaces program is extremely
hard. The alternative could be to find new algorithms, in order to evade
the long auto-correlation times that have characterized our
simulations so far.

Even if no such transition exists for finite $\lambda$, one could
still attempt to study a continuum theory in the strong coupling
limit, as is done for QCD, for instance.  To do so, we would like to
examine global quantities, such as masses extracted from normal-normal
correlation functions, rather than just the local quantities (energy,
e.g.)  that we have measured.  Measuring these correlations requires a
definition of distance on these triangulated lattices; the most
successful definition of the metric is based on the propagation of
massive particles (via inversion of the Laplacian) on these lattices
\cite{MigLap}.

\subsection{Importance to Other Fields (Biological Membrane, Crystalline Membrane, bilayer etc)}

Maybe I need a section about this??


%\include{sec5}
% $Id: sec5.tex,v 1.3 1994/06/25 13:44:06 han Exp $

\newpage

\section{Aspects in Computer Science}
As we have seen in the previous chapters, Monte Carlo simulations of
random surfaces require enormous computer power. The use of high
performance computers is central to our goal of understanding random
geomtry in many guises. With the most recent advances in the field of
supercomputing and massively parallel computing, computation can thus
lead to a vastly improved understanding of the fundamental
interactions of nature from gauge theories of the strong and
electroweak interaction to the most elusive interaction in the quantum
domain fo far - gravity. 

However, with dramatic changes in technology ahead, we have to ask
ourself the following question: how do we approach the problem of
high-performance architecture design and high-performance engineering
and scientific computing? For example, the new technology makes
feasible massive parallelism. How much additional effort should be
invested in increasing the performance on a single processor before we
seek higher levels of performance on multiple processors?  There are
no simple answer to these questions. The latter is always essential to
the first. We need a combination of solutions, and what we choose
almost certainly will be application dependent, since at this stage we
have not constructed a general machine that would be equally effective
for all high performance applications yet.

In the past, we have seen many different techniques used in hardware
to improve performance for individual processor, such things as
instruction buffers, cache memories, pipelined execution and RISC
computer architecture~\cite{STONE,Stallings}, have appeared in many
commercial machine implementations. And people created parallel
computers to overcome the current limit of technology in order to
achieve even faster speed. However, can a application well and easily
utilize these hardware capabilities?  This is always a complicated
story. In terms of software issues, it has to address to both the
level of compiler design and the level of user's applications and
algorithms. As computer clock speed keeps increasing, the issue of how
well applications can utilize a computer's cycle becomes even more
important, since many numerical applications are so demanding of
computational cycles and call for sophisticated floating-point
processors in the architecture~\cite{FOX}. And the parallel computers
bring even bigger challenge to software design and calls for
innovations of completely new algorithms. 

We also note that the power of graphics system with the increeasing
power of high performance computers has been increasing and made it
possible for the scientific and engineering communities to gain new
insight into their disciplines. Both the size of scientific problems
and the quantity of data generated presenta large challenge to us:
namely how to understand the results of their computations. Solving
these problem interfactively is essential and requires significant
computation and graphics power. 

\subsection{Challenges of the Random Surfaces to Computation}

The computer science challenges for computational intensive and data
stucture complex application like random surface simulations are
centered on how to improve individual processor's performance and how
to decompose the problem and map it onto parallel processors. 

The original code we used was written by Baillie, Johnston and
Williams~\cite{DIMECODE}, and was based on Distributed Irregular Mesh
Environment(DIME), a general software package for handling dynamically
triangulated meshes~\cite{DIME}.  Dynamic unstructured meshes of this
type are also used for finite element simulations and computational
fluid dynamics.  This code was written in C, with the mesh represented
as a linked list.  The generality of the code meant that it used a lot
of unnecessarily complicated data structures, which increased the
memory requirements and decreased the efficiency of the program.

In order to improve the performance of the code, we rewrote the
program from scratch without using DIME. The new program was written
in Fortran, in order to make it easier to parallelize by using Fortran
90, Fortran D, or CM Fortran. The new code was simpler and more
specialized.  Consequently it ran approximately 8 times faster than
the previous code on the IBM RS/6000.

However this new code still ran very slowly on certain machines,
especially the Intel Touchstone Delta, which uses Intel $i860$
processors. For this code we realized only about 1 MFlop on the
$i860$, which is theoretically an 80 MFlop processor for 32-bit
operations. 

We believed that we could substantially improve the performance of the
program on the $i860$ and the IBM RS/6000 by careful optimization of
the code, and in particular by making better use of the data cache.  A
floating point operation can be done in a single cycle on modern RISC
processors, however access to data not in the cache may take many
cycles. To get near optimal performance from these processors we
therefore need to be very careful to structure the code in such a way
as to make best use of the data cache, by keeping data as local as
possible.  This is a non-trivial problem for applications with dynamic
data structures such as the simulation of dynamically triangulated
random surfaces.  This is very similar to problems of data locality on
parallel processors, where there is an even greater penalty in
accessing data which is non-local.  For programs such as parallel
random surface simulation, dynamic domain decomposition is required
for both load balancing and efficient access to data. A similar
approach can also be applied to optimize sequential programs on
processors with a hierarchical memory structure.

Unstructured numerical applications~\cite{DAS,DIME} have been one of
the most computationally demanding areas and have the most complicated
but the richest data structures. Naturally people like to use them to
evaluate the overall performance of a computer system in dealing
irregular problems for both hardware and software
aspects~\cite{DAS,TOM}.

The Monte Carlo simulation of dynamically triangulated random surfaces
happens to be one of the most suitable application to be investigated
for state-of-the-art computers. It has very irregular with complicated
data structures ({\it Figure} 1) and demands very intensive floating point
calculations. The computation is very time consuming. For example,
running on HP 9000 24 hours a day, the computation takes about 3
months to have one statistically meaningful physical data point for a
mesh with 1152 nodes, even though our code is the fastest sequential
code for dynamical random surface simulation.

We had done the largest simulations and most accurate measurements of
physically interesting observables so far on 2-d systems with 2304
nodes. This gave us more meaningful physical results. However,
correlation length for most of observables is a major problem. The
simulation takes hundreds of thousands of sweeps of updating the mesh
to yield one statistically independent configuration. The existence
order and critical exponents of this {\it crumpling transition} are
still uncertain. That is our driving force to speed up the code by
optimizing it, looking new algorithm and using parallel computer. The
study of computational behavior of the random surface simulation on
high performance computer provides a basis for on-going High
Performance Fortran compiler support~\cite{TOM}.

\subsection{The Random Surfaces Fortran Program}

The program used for the Monte Carlo simulation of string theories was
written in Fortran. The following outlines the crucial part of main
program.

\begin{verbatim}
	program main-outline

	CALL mesh-set-up
	CALL global-data-initialization

	DO i = 1, n_measurements
	  DO j = 1, n_sweeps
	    CALL update-nodes
	    CALL update-links
	  ENDDO
	  CALL measurements
	ENDDO

	END
\end{verbatim}

The program first constructs a initial mesh. {\it Figure} 4 shows a initial
mesh with torus topology. We can see that one sweep consists of two
parts : {\it update-nodes} and {\it update-links} . The subroutine
{\it update-nodes} is used to update each node's position of mesh in
the three dimensional space ({\it Figure} 5). The subroutine {\it
update-links} is then used to change the connectivity of the mesh,
i.e., to try to flip each link between nodes in the mesh ({\it Figure} 6).
The Metropolis algorithm is used to update both the nodes and links of
the surface. As the simulation proceeds, the geometrical data and the
physical observable associated with nodes and edges have to be
recalculated.

The node update part loops over all nodes. In the loop it randomly
picks an attempted position for the current node and calculates the
energy change in the simulated system between the old position and the
attempted position ({\it Figure} 5). The Metropolis algorithm
determines if the attempted position should really be used to update
the node's position or not. The energy calculation is based on the
length of the edges and the so called edge action, i.e. the dotproduct
of two normal direction vectors of two triangles along the common
edge. This dotproduct is actually calculated in terms of the two
altitudes in the triangles along the common edge. The edge action
calculation is very expensive since the calculation not only involves
the data associated with all nearest neighbors but also the next
nearest neighbors ({\it Figure} 7). However the node update is still
less irregular in the sense that the connectivity is not changed.
Therefore the data retrieving process from cache or main memory for
the calculation of node update is still relatively smooth. It probably
does not experience very frequent interruptions or jumps since
hopefully most of the required data are laid out in memory
consecutively, so useful data may stay in the cache for a while before
moved out.  With fixed connectivity, there are fewer branchings and
fewer pointers to cause the memory jumps than in subroutine {\it
update-links} where the connectivity is changing as each link flips.
Therefore {\it update-nodes} produces fewer cache miss and page fault
than {\it update-links} .  However, even in the case of update node,
the mesh is still irregular.  Each node may have different number of
neighbors, and therefore a different number of edges. These
complexities may cause load balance problems for later parallel
computation.

The subroutine {\it update-links} is used to randomly pick a link and
attempt to flip it to another diagonal as in {\it Figure} 6. The calculation
of the energy change between the attempted configuration and old
configuration is still similar to the one in {\it update-nodes} which
uses the Metropolis algorithm. However at this time, not only the
calculation is as expensive as the {\it update-nodes} case, but also
the data structure layout in the memory associated with the nodes and
edges may become very irregular. To calculate observables, we may have
to look for them throughout a big area of memory since the pointers
may point to different places as we add links to one node and remove
links from others. The data associated with the changing edges are
scattered all over the memory and logically closely related data
information are moving far apart in memory as the links flip.  Data
locations are less predictable (This will be explained more later in
the paper). Therefore, as the simulation proceeds, the data
associated with the changing edge may be moved in and out from the
cache more frequently, and the data required to calculate observables
may be distributed over different pages in virtual memory space.
Therefore the behavior of {\it update-links} part may appear to be
more "violent" in terms of memory access and becomes more interesting.

All of these characteristics are very good features to be used as a
probe to investigate the performance of application software, compiler
and memory hierarchy.  This is because one code has two typical modes
of behavior. By monitoring the performance of the two parts, we may
gain some insights into many computational issues.

\subsection{Optimization of the Random Surface Code for RISC}

\noindent

(1) Optimization and Memory Hierarchy
\medskip

Our Fortran code for the random surface simulations ran very slowly on
the Intel $i860$, and we were convinced that we could improve the speed
of the code substantially for this and other modern RISC processors
by some careful optimization.

We believed that the poor performance of the program on the $i860$ processor
was probably due mainly to cache misses,
since the off-cache memory access for this processor takes many cycles,
whereas a floating point operation can be done in a single cycle.
This is the case for most RISC processors, and can be a problem in
utilizing these processors efficiently.
We therefore attempted to understand why the program ran so slowly,
and to try to optimize the program for RISC processors such as the $i860$
and the RS/6000 by investigating possible problems, in particular memory
access.

Note that this is also an essential step in optimizing parallel
programs, since we want to distribute data over processors in such a way
that the memory accessed by any processor is as local as possible, to
minimize communication costs between processors. On a sequential machine
we have a similar kind of memory hierarchy, with on-cache memory
corresponding to on-processor memory for a parallel machine,
and off-cache memory corresponding
to memory on a different processor, which requires extra time to access.

Based on the idea of memory having a hierarchy structure, and the
particular characteristics of certain RISC chips, we can outline a fairly
systematic method for optimizing the program. These methods are quite
general, and can be applied to any numerically intensive program run on
a fast RISC processor.
We have benchmarked the results of these optimization techniques on the
Intel $i860$, IBM RS/6000, HP 7000, DECstation 5000, and Sun SPARC 1+.


\bigskip
\noindent
(2) Optimizing Compilers and Basic Optimization Techniques
\medskip

We first investigate what the compiler provides. Optimization flags with
compilers provide local and global optimizations,
and may perform some pipelining. A comparison of the speed of the
program before and after compiler optimization for all the different
machines is shown in Table~\ref{TABSTRING}.

For the $i860$, we found that the speed is increased by almost a factor of
3 just by using the NOIEEE compiler flag. This causes float and double
divides, which are otherwise extremely slow since they are done in software,
to be done using an inline divide algorithm. This gives substantial speed-up
of functions such as division, square root, exponential and arccos which are
used in the strings program. In a similar vein, the RS/6000 Fortran compiler
has a flag (qrndsngl) which needs to be turned on in order to ensure strict
adherence to the IEEE arithmetic standard (i.e. IEEE standard arithmetic
is {\it not} the default for this compiler).
The strings code runs about 10\% slower with this flag set.

We use our own random number generator subroutine many times in the inner
loop of the program, and each call to this subroutine has a function call
stack overhead.
The IBM and Intel compilers provide an inlining compiler flag, which allow
calls to specified functions, or functions of less than a certain size,
to be placed into the code (inlined) rather than using a function call.
By using the inline operation on the random number generator we get about
a $4\%$ speed-up for the $i860$, but no noticeable speed-up for the RS/6000.
The same results are obtained by manual inlining.
There is a trade-off here, in that bad inline effects
may actually decrease the performance of the code, by
increasing the code size and thus
increasing overhead due to the small size of the instruction cache.
Compilers for other machines may do automatic inlining at a certain level of
compiler optimization, and one should be careful to check that this does
in fact increase and not decrease the performance of the program.

We know that current compiler technology has not yet matured to the point
of automatically taking advantage of many of the features of modern RISC
chips. Thus the user needs to carefully reconstruct the program and fully
expose the availability of pipelining and data locality to the compiler,
in order to take maximum advantage of the data cache.

Firstly, we must try to keep the most frequently used variables in the
cache, and reuse them as much as possible while they are still there.
This may require re-ordering parts of the code.
We also tried to reduce unnecessarily large array sizes so that all, or at
least most, of the working set data is in the cache.
Thus after the transient period for loading this data onto the cache,
most of the memory access is to fast cache memory.
Even better is to keep data which is re-used in registers.
This can be done explicitly by the programmer in C by using the {\it register}
variable type. In Fortran we need to allocate such data to temporary locally
declared variables, and hope the compiler takes advantage of this.

Secondly, in trying to optimize code, one usually thinks of an assignment
as being fast and a multiplication as being slow, however this is not the
case with modern RISC processors.
On the $i860$, for example, a load operation takes two clock cycles
and a store operation three cycles while add or multiply take just one.
We should thus eliminate redundant assignments and minimize memory access,
use constants rather than variables, and try to avoid data conversion.

Thirdly, we try to minimize the number of expensive calculations, such
as division, square root, exponential and arccos, which all occur in the
strings code.
If a calculation is expensive and repeatedly used in several places,
it may be possible to calculate it once and assign it to a global variable,
so further calculations are reduced to using this ``look-up table''.
We should also try to utilize the concurrent add and multiply operations.
An addition combined with a multiplication will be performed in about the
same time as a multiplication on most RISC processors.

Fourthly, branches (IF statements) and loops (DO or FOR statements) can
have a large overhead.
We should try to avoid IF statement since they slow down any pipelining.
Optimizing loops is somewhat problematic.
Using the DO loop is a very good way to utilize the instruction cache,
since it allows a lot of code to reside in the generally small instruction
cache space. In most processors the CPU can take care of the loop while
the floating point unit deals with the operations in the loop.
However there is still a substantial overhead in using a DO loop,
so we should merge DO loops as much as possible, and in cases where the
loop contains very few instructions it should be ``unrolled''. This means
that if the number of iterations of the loop is very small (for example,
a loop over 3 dimensions), the loop should be eliminated and the instructions
written out explicitly for each iteration (each dimension).
If the number of iterations of the loop is large, say 100 iterations
of a single line of code, then it should be unrolled into a loop
over say 5 iterations of 20 lines of code. The optimal number of lines
of code in the loop after such a split should be such that the loop just
fits in the instruction cache. unrolling of loops should be done by the
compiler, but this is not always the case.

Also, since data is usually loaded into the cache a few words at a time,
we should
try to keep the loop stride smaller than the length of this cache line to
avoid possible ``thrashing'' of the cache (continual reading and writing
of data to and from cache), and also increase the chance of pipelining
multiple read and write operations since the working data will be in same
cache line or nearby.

Another crucial point to note for multiple loops over elements of arrays
of more than one dimension is that the loops must be ordered in such a way
that the inner loop corresponds to the array index which changes fastest.
In Fortran this is the first index, while in C it is the last index, so
optimal loop ordering will depend on the language used.


\bigskip
\noindent
(3) Program-Specific User Optimization
\medskip

After implementing the generic optimization techniques given above, we
turned to more problem-specific methods to speed up the code. Here we
are still interested in minimizing memory accesses and maximizing cache
usage, but now we are aiming to do this by studying the specifics of
our particular problem,
and finding ways to change the algorithm so as to better match the data
structures and data access to the hierarchical cache memory of the
processor.
By appropriately ordering the operations at each mesh point, and in
particular the traversal of the mesh data structure, so as to maximize
data locality, we realized quite substantial gains in performance.

Here we show a typical example. Initially, each node in the mesh has six
nearest neighbor nodes, so it has six connecting links. Each link has a
pointer $node\_of\_link$ to point to its node and a pointer
$link\_next\_link$ to point to its neighbor links. Initially
everything is in order (Fig. 29).
As the mesh is changed during the simulation by flipping the links,
the associativity between nodes and links and neighboring links changes.
Links pointing to the same node may now not be located in contiguous
sections of the memory, and the link pointer may not point to the nearest
neighbor link among the links around a node (Fig. 30).
That is the case in the original version of the code, for which the links
around a node are not ordered after flipping.
However, the calculation of physical quantities
associated with each link requires the information at the nearest
neighbor links around their boundary nodes. Since the links are not
ordered, to figure out the left side link and right side link
of a given link, the original code first looks for the left side and right
side nodes $S_1$ and $S_2$ (Fig. 29), and then makes a comparison to
see which link connects the node pair ($S_0$,$S_1$) and ($S_0$,$S_2$).

\begin{verbatim}
       ...
       s1 = node_left_to_link(this_link)
       s2 = node_right_to_link(this_link)
  111  temp_link = start_link_of_node(s0)
  112   if (node_of_link(temp_link) .eq. s2) then
           s0s1 = temp_link
           goto 113
        else
           temp_link = link_next_link(temp_link)
           goto 112
        end if
  113   continue
        ...
\end{verbatim}

Here we have changed the algorithm to improve data locality.
At the time we flip the link, we re-order the links immediately, each link
getting a pointer to its nearest neighbor link (Fig. 31). So we get
its nearest neighbor links ($S_0S_1, S_0S_2,$ etc.) basically for free.

\begin{verbatim}
        ...
        s0s1 = link_next_link(this_link)
        s0s2 = link_last_link(this_link)
        ...
\end{verbatim}

By implementing this and making other similar changes, we achieved the
biggest improvement in speed. It eliminates a lot of IF statements, pointer
chain searching operations, and several unnecessary large arrays.
It also provides much useful information for other calculations at no
extra cost.

In general, it is much harder to exploit data locality in a dynamic
mesh than a fixed mesh. Because the neighbors of nodes and links and their
corresponding ordering are changing as the simulation continues, the
neighboring physical memory locations may not reflect the neighbor of a
link or a node, and vice versa (Fig. 30). The optimization may require a
run-time preprocessing strategy to predict more precisely the data locality.

\bigskip
\noindent
(4) Performance Comparisons
\medskip

We have compared the performance of this code and the effect of these
optimization techniques on a number of current RISC-based processors and
workstations, in particular
the Intel $i860$, IBM RS/6000, HP 7000, DECstation 5000, and Sun SPARC 1+.
Table~\ref{TABSTRING} in section 3.2.4
shows the performance of the program on all these
different machines for various system sizes.
We show timings for the code before any optimization, with compiler
optimization, and with optimization by the user involving rewriting parts
of the code as outlined above.

\begin{table}
\begin{center}
\begin{tabular}{|c|c|c|} \hline
 & & \\
& \multicolumn{1}{c|}{Unoptimized} & \multicolumn{1}{c|}{\quad Optimized \quad} \\
&                       code        &                             code \\
                 &              &           \\ \hline
                 &              &           \\
IBM RS/6000-550  &   3.8        &    3.7    \\
                 &   6.9        &    8.2    \\
                 &              &           \\
HP 7000          &   4.1        &    3.5    \\
                 &  11.7        &   12.1    \\
                 &              &           \\
Intel i860       &   1.5        &    3.0    \\
                 &   2.8        &    4.7    \\
                 &              &           \\
DEC 5000         &   2.5        &    2.6    \\
                 &   2.5        &    2.6    \\
                 &              &           \\
Sun SPARC 1+     &   1.0        &    1.0    \\
                 &   1.3        &    1.5    \\
                 &              &           \\ \hline
\end{tabular}
\bigskip
\end{center}
\protect\caption{
Comparison in performance for random surface strings code
on a number of different workstations. These are normalized so that
the slowest is 1. It is difficult to obtain an exact flop count for
this code, due to the use of intrinsic functions, however the numbers
given here are also very close (probably within 10\%) to the
MFlop rate for this code.
The two values presented are for Fortran codes with (bottom) and without
(top) compiler optimization.
\protect\label{TABSTRING}}
\end{table}

In Table~\ref{TABSTRING}.  We show timings for the code before any
optimization, with compiler optimization, and with optimization by the
user involving rewriting parts of the code as outlined above.  It is
interesting to note that the slower processors (the DEC and Sun
machines) gain relatively little from these optimizations or compiler
optimization, whereas the more powerful processors (IBM and HP) show
substantial improvement. The biggest improvement is found for the
$i860$, which is most likely due to the complexity of its architecture
and the consequent difficulty in creating good compilers for this
processor.

\bigskip
\noindent
(5) Conclusions
\medskip

>From the study of irregular dynamical mesh program optimization, we
have learned that we need to be careful to construct the code and the
data structures so that they make most efficient use of the memory
hierarchy characteristics and pipelining of modern RISC processors.
Current compilers can provide good optimization, but do not recognize
and exploit all the information on data locality and pipelining
opportunities. Substantial gains in performance can be achieved if the
user is prepared to restructure the code to help provide the compiler
with this information, especially programs with adaptive, irregular
data structures such as the random surfaces application.

Our results show that the dynamical part of the program needs to have
more aggressive code re-arrangement or even algorithm modification in
order to achieve a relatively big performance improvement. Simply
using the basic general optimization techniques does not have real
effect on the dynamical part although it does speed up the static
part. That means the compiler needs to be more intelligent so that it
can look deeper into the program and extract more information about
data association if the compiler wants to do this level optimization.
That is one issue that High Performance Fortran will eventually
address. Dynamical run-time techniques have to be used to explore more
precisely the behavior of data locality. we are trying to extract more
general strategies from ours and other similar type of irregular
applications to support the High Performance Fortran development.
This study is a pre-investigation for the on-going development of the
High Performance Fortran compiler by Syracuse University, Rice
University and University of Maryland.

The approaches we used here can be applied to improve the performance
on almost all major types of RISC processors.  These data locality
techniques can also be used by data parallel compilers such as those
for High Performance Fortran to allow these languages to outperform
conventional Fortran even on sequential processors. Thus the High
Performance Fortran not only achieves the parallelism but also
exploits the capacity of each individual processor, especially for
very irregular problems.

\subsection{Parallelization of Random Surafces Program --
on Distributed Computing, SIMD \& MIMD}

\subsubsection{Independent Parallelsm}

Monte Carlo simulation approximates a large or infinite sum or
integral over all possible states of the system by a finite sum over
representative states of the system (usually called {\it
configurations}). This is a statistical process, with a corresponding
statistical error which is proportional to the square root of the
number of independent states sampled.  Monte Carlo simulations
therefore have a large amount of trivial parallelism, in that
different processors can be generating different configurations
completely independently, by using different starting configurations
and different random numbers. These independent results can then be
combined to reduce the statistical error in the simulation.  As long
as the system size is small enough so that each processor can hold all
the data for a simulation, this procedure will give perfect speedup
over a MIMD parallel computer or an array of workstations.

We have used this method of trivial parallelization running
independent random surface simulations on networks of workstations and
on MIMD parallel computers such as the Intel Touchstone Delta and the
nCUBE/2.  These runs used different parameter values or different
meshes with different random numbers for each simulation.  This kind
of ``job level'' parallelism is common to many kinds of simulations in
science and engineering. Usually simulation involves studying changes
in the system as parameters are varied. Sometimes the choice of new
parameters will depend on the results of the simulation with the
current parameters, so this is a sequential process. However in many
cases one would like to know results for a large number of parameter
values which can all be run independently.

This approach is ideally suited to very coarse-grained parallel
machines, and especially for distributed computing over networks of
workstations.  There are at least three methods of invoking job level
parallelism on distributed networks, which we discuss below. We have
done our physics production runs with the strings program using the
first two methods, neither of which we found to be satisfactory.  This
has led us to think of programming paradigms and systems software
which would be ideal for this type of physics application, and this is
what we propose as the third method.

\bigskip
\noindent
(1) The Brute Force Approach
\medskip

The simplest but most tedious method is just to log on to any
available machines in the network and submit a different job to each
of these machines. This method is too time consuming to use on any
more than a handful of workstations.  It would be possible to set up a
shell program to remotely run these jobs on different processors,
however deciding which machines to use generally requires checking
that the machine is not already in use.  Physicists and other
scientists who might wish to use networked workstations in this way
usually do not have the systems programming knowledge to implement
such a method.

\bigskip
\noindent
(2) Portable Parallel Software
\medskip

Job level parallelism can be used on distributed computing networks
and MIMD parallel computers using parallel software.  It is easy to
set up parallel code to do this, since only data input and output need
to be managed, since all computation is sequential and requires no
communication.  The problem with this method is that although the
programs are run independently, they generally require synchronization
every time data is input or output.  If results are output frequently,
this communication may cause unacceptable performance degradation.
Synchronization overhead can be reduced by altering the code slightly
to buffer data on each processor and thus output larger blocks of data
less frequently.  This was not required for the strings code, which
performed at virtually 100\% efficiency when distributed across the
RS/6000 network, even for the smallest system sizes used.

Synchronization during data output can also lead to performance
problems if one or more of the processors in the network is slower
than the others, perhaps because it is also running other jobs.  This
can be avoided by writing the data from each processor into separate
files which can be accessed asynchronously. This can be done in
Express, for example. This approach allows all the processors to run
completely independently.

We still have the problem of ``load balancing'', in the sense that
some machines in the network may be loaded with other jobs, and thus
take much longer to finish their work. It would be helpful if there
were some kind of dynamic load balancing functionality, whereby if a
machine became loaded with another job, then the process being run
under Express would be migrated to another unloaded machine in the
network, if one was available.


\bigskip
\noindent
(3) A Transparent Network
\medskip

It would be preferable to be able to run multiple jobs over the
network without having to deal with parallel software (such as
Express) at all.  Ideally, one would like to be able to just set up a
standard Unix runfile which submitted multiple jobs, with different
input and output files, to the {\it network}, rather than to a
particular workstation. Some clever systems software would then
distribute these jobs to whichever machines on the network had the
smallest load and were not being used interactively.  Again, some load
balancing software would be required, so that if an interactive
session was started on a workstation, the job would be migrated to
another workstation. This would all be transparent to the user, who
would just submit jobs to the network and not care where they were
run.

This works very well for relatively small systems, however for large
systems we strike a snag related to the problem of critical slowing
down, which we have seen is a major problem for Monte Carlo
simulations of spin models near phase transitions. Random surface
simulations also suffer badly from critical slowing down. A
consequence of this is that the {\it equilibration time} grows rapidly
with the size of the system. This is the time required for the Monte
Carlo algorithm to move the system from the initial state, which can
be any configuration, to an uncorrelated state representative of the
system being simulated, i.e. with a particular temperature or system
parameter (in the case of string theories, this is the extrinsic
curvature parameter $\lambda$). For trivial parallelism, each
processor has to run for at least the equilibration time before it can
start producing usable data, so this is effectively wasted time. For
large systems, this time may be so great that in practice it is not
feasible to equilibrate the system on a single processor, whereas a
system being simulated in parallel over many processors may be
equilibrated and produce useful data in a reasonable time period.

\subsubsection{Parallel Random Surfaces}

Have seen the problem with independent parallelsm in the previous
section, we also started to work on parallelizing the random surface
code on SIMD machines such as the CM-2 and MIMD machines such as the
CM-5, the Intel Delta, and the new IBM RISC-based parallel machine.
Since this application involves a dynamic irregular mesh, we would
expect that SIMD architectures would fare poorly on this problem.
However they may do reasonably well on problems with fixed
triangulated surfaces which do not change during the simulation.

Efficient parallelization of a dynamically triangulated random surface
is a very difficult problem even on MIMD machines, since dynamic load
balancing will be required.  This is because the mesh changes
throughout the simulation, so that the neighbors of a point, which
will originally reside on the same processor or a neighboring
processor using standard domain decomposition, will change so that
they may end up on distant processors, thus increasing the
communication cost.  Also the irregularity of the mesh means that it
is not simple to determine which points can be updated in parallel, as
it is for the regular fixed lattices we use for the simulation of spin
models.

% parallel coloring part
In DTRS simulations, both the vertex and edge update operations are
generally done using the Metropolis Monte Carlo
algorithm~\cite{baillie,binder,gould}.  The update of the variables on
each vertex depends only on the values at neighboring
vertices.\footnote{In general this will also depend on next-nearest
neighbor vertices, but for simplicity we will consider only the
nearest neighbor problem. The general case is a simple extension of
this problem.} The Metropolis algorithm requires that dependent
(neighboring in this case) values cannot be updated simultaneously
(i.e. in parallel).

If we describe the mesh in terms of an undirected graph with vertices
and edges, then the Monte Carlo update of the mesh is achieved using
two basic operations~\cite{baillie}: the update (or ``flip'') of an
edge of the graph, which changes the structure of the graph (see
Figure~\ref{flip}); and the update of the variables associated with a
vertex of the graph, which, for example, might be the position of the
vertex in the embedding space, in which case the update is just a
``move'' of the vertex (see Figure~\ref{move}).  These two updates are
sufficient to generate all possible triangulated
meshes~\cite{baillie,bowick}.

This is the well-known {\em graph coloring} problem.
An {\em undirected graph} G is a set of vertices $V$ and a set of edges $E$.
The edges are of the form $(i, j)$ where $i, j \epsilon V$.
% We shall only consider graphs where $i \not= j$ and a particular edge
% $(i, j)$ may appear at most once in $E$. 
A {\em coloring} of a graph $G$ is a mapping ${c:V \rightarrow
\{1,2,\ldots,s\}}$ such that $c(i) \not= c(j)$ for all edges $(i, j)
\epsilon E$.  $c(i)$ is referred to as the {\em color} of vertex $i$.
Vertices $i$ and $j$ are said to be {\em neighbors} if $(i,j) \epsilon
E$.  The number of vertices is denoted by $V$.

Note that for the DTRS simulations, we are updating both the vertices
and the edges, so for a parallel implementation we need a vertex
coloring and an edge coloring.  In this paper we will concentrate
solely on the problem of coloring the vertices of a graph.

Many simulations in computational science discretize the continuum
world as a triangulated mesh. This is particularly common for partial
differential equation (PDE) solvers~\cite{DIME,pde2}.  These meshes
are irregular and often adaptive, that is, they change during the
course of the simulation~\cite{adaptive}.  For example, in
computational fluid dynamics, the simulation of airflow over an
airplane would use a finer mesh in turbulent regions and a coarser
mesh in regions of laminar flow. The positions of these regions are
not known initially, but must be identified during the course of the
simulation. 

Here we introduce some parallel graph coloring algorithms based on
well-known sequential heuristic algorithms, and compare them with some
existing parallel algorithms.  These algorithms are implemented on
both SIMD and MIMD parallel architectures and tested for speed,
efficiency, and quality (the average number of colors required) for
coloring random triangulated meshes and graphs from sparse matrix
problems.

Since this problem is so challenging to parallelize, we expect that
the implementation of a parallel algorithm for this dynamic irregular
mesh problem will also provide input for new parallel languages and
compilers, such as Fortran D and High Performance Fortran.

We would like to speed up the graph coloring part of these algorithms
by doing the coloring in parallel.  However our goal is to reduce the
run-time for the whole computation.  There is a trade-off here between
the time spent in coloring the graph and in updating the edges and
vertices of the graph.  For an adaptive grid PDE solver, many updates
will generally occur between adaptive refinements of the graph which
require a new graph coloring. In this case the percentage of the time
spent in coloring the graph is very small, so it is worth spending
more time to get a better coloring, which should improve the
parallelism and reduce the update time.  However for a DTRS
simulation, every iteration involves an edge update, which changes the
structure of the underlying graph, so the graph must be re-colored
after every iteration.  The graph coloring could therefore provide a
substantial overhead unless it is much faster than the update time.
In this case we are mainly interested in speed, and may be willing to
make do with a good coloring, rather than a better coloring which
takes much longer. We are therefore interested in studying and
comparing a variety of parallel graph coloring algorithms.

For most of the applications that requires graph coloring, finding a
good graph coloring with a small number of colors is only part of the
problem. We must also be concerned with load balancing, since each
coloring is followed by an update step in which the variables
associated with vertices of the same color are updated in parallel.
Thus we need also to ensure that the number of vertices of each color
on every processor is approximately the same, in order to obtain good
load balance in the update step. Our goal is therefore not just to
obtain a good coloring, but to obtain a {\it balanced} coloring, that
is, to minimize the number of colors required taking into account the
load balance constraint, that the number of vertices of each color be
approximately the same on every processor. It may be advantageous to
make do with a larger number of colors if this makes the load more
evenly balanced.  

We will concentrate on the first of these two goals,
{\it optimal} graph coloring and suggest how the algorithms presented
may be modified to achieve {\it balanced} graph coloring.

I did not have enough time to continue this investigation. Several
other people in NPAC went on and implemented the parallization of
random surfaces, using the action with only Guassian term. They found
The LDF algorithm appears to perform exceedingly well in both
architectures.  The processing time required remains lower compared to
many of the other algorithms even as the size of the problem grows
larger.  In particular the SDL and MIS algorithms require much more
communication in each pass, which would account for their poor
performance at large problem sizes and with more parallel processors.
Even noting that the communication required by the LDF algorithm is
equivalent to that required by the J--P algorithm, it consistently
performs better.  The LDF algorithm consistently achieves between 5
and 6 colors on planar graphs, which is excellent considering the
NP--Hard nature of the optimal 4-coloring.  The LDF algorithm is
fairly easy to implement and to understand, which makes it relatively
easy to incorporate into many parallel architecture codes that may be
written at some point in the future.

For some algorithms, it may be important that there is no bias towards
updating certain vertices before others. This is known as the
requirement for {\em detailed balance} \cite{binder,gould}. For
example, the LDF and SDL algorithms will both tend to color
large-degree vertices first, assigning them as a small number
``color''.  If the vertices are updated in color order, then there
will be a bias towards updating these vertices first. This problem can
be avoided by picking the color sets to be updated in random order.
With the Jones-Plassmann and MIS algorithms, this problem does not
arise because there is no bias towards a particular set of vertices.

For applications such as DTRS, which require regular re-coloring of
the graph, the LDF algorithm is probably the best to use, since it
takes the same time as J-P, but requires fewer colors. For
applications such as PDEs, the SDL algorithm, which takes longer but
uses fewer colors, may well be preferable since the coloring is
performed once only.

A further refinement of the coloring algorithm is a balanced coloring,
requiring that the roughly equal numbers of vertices have each color.
Having a small number of vertices of any one color means that there is
not much parallelism to be exploited in the update step for the
parallel application (such as a PDE solver or random surface
simulation) that is using the results of the graph coloring.  A
balanced distribution of colors makes it easier to load balance the
work of updating and effectively exploit a parallel machine.  A simple
modification to the sequential versions of the algorithms described
here will achieve a balance of colors. Instead of picking the smallest
available color, one picks the least used of all available colors
\cite{huang}.  This idea can be adapted readily to a MIMD
implementation; each processor chooses from the least used color in
its local patch of vertices. In a SIMD implementation this is not
possible because each vertex knows only the colors of its neighbors.
An alternative strategy in this case would be to pick a color at
random from the set of legal colors for a vertex.  This could result
in a slight increase in the average number of colors required, however
this cost should be outweighed by the improved load balance.  No
experiments with color-balancing were performed.

And they also disappointedly found no speedup with the number of processors
used on the iPSC/860. One reason for this may be that the algorithms
as coded sent the set of weights for the whole graph to every
processor and the set of colors for the whole graph to every
processor. 

Much pioneer work has been done in parallelization of random surfaces.
However, we still have lots of more work to do. We have not had a
parallel random surface parallel program. If we add in the extrinsic
curvature term in the action in the parallel program, things would
becomre even more complex.

\subsection{Visualization for Monte Carlo Simulation (especially 3D)}

Graphics and visualization are increasingly becoming indispensable
tools for analyzing and understanding large data sets and complex
physical problems. Many of our physics applications involve complex
and irregular data structures, and visualization tools are extremely
useful for analyzing and understanding the physical system and the
complex algorithms used to simulate the system. Graphical tools can
also be a useful aid in debugging such algorithms.  The IBM RS/6000
workstation provides a powerful graphics capability which we have used
extensively in our studies of physics applications and parallel
algorithms.

The machine with 24-bit color also is licensed for the Advanced
Visualization System (AVS) from Stardent and Mathematica from Wolfram
Research.
	
Graphics and visualization are extremely useful in the study of Monte Carlo
simulation of spin models, on many different levels. They can aid the
understanding of the physical system, the algorithms used to study the
physical system, and the implementation and debugging of these algorithms.

% \bigskip
\noindent
(1) Visualization Using AVS
\medskip

Two dimensional dynamically triangulated random surfaces consisting of
hundreds of points embedded in three dimensional space-time are very
difficult to picture without some sophisticated visualization tools.
For our research, we are particularly interested in seeing how the
structure of the surface changes from smooth to crumpled as we change
the curvature parameter, and the variation in shape of the surfaces which
are typical of particular parameter values.
%
We also wish to see how rapidly the surface changes during the simulation,
and how certain structures, such as large spikes, can be created and
take a long time to disappear, which contributes to the problem of
critical slowing down. New Monte Carlo algorithms which can reduce critical
slowing down in random surface simulations is an area we have been studying,
and plan to investigate more in the future. As we have seen with spin models,
visualization is extremely useful for the study of such algorithms.

Each 2 dimensional triangular cells cell is defined by vertices and
connecting edges. We have used the Application Visualization System
(AVS), since is particularly powerful at handling 3D images,
especially for unstructured data. It also has a very simple and
powerful user interface, which allows a complex visualization program
to be constructed merely by connecting separate modules using an
interactive graphical environment. AVS also allows the user to
interactively adjust lighting, shading, coloring, and orientation of
the image.  Another beneficial feature is that it is available for
many different systems, including Silicon Graphics, Sun, DEC and IBM
workstations.

We have written an interface module to read in a file containing the random
surface data, and convert it into a format which is understandable by AVS.
AVS then does the rendering and display of the surface.
Rendering, rotating, and zooming this complex 3-D image under AVS running
on Sun or DEC platforms can be very slow, however with the IBM RS/6000
these operations are done very quickly, especially when special graphics
hardware is utilized.

Figs. 21-25 show examples of dynamically triangulated random surfaces
with 576 points displayed using AVS.
%
% The colors represent the curvature of the surface, with blue corresponding
% to smooth sections of the surface, and red representing sharply peaked
% elements. Concave sections of the surface are shown by the addition of
% green to the color of a vertex, giving yellow, green, or orange hues.
%
We are still experimenting with different color maps to reflect
different properties of the surface, such as curvature, convexity, etc.
Here we have just used random colors at each node.
We have used Guraud shading, which interpolates the color between nodes of
the surface.

Fig. 21 shows a wire-frame rendering of a small initial configuration in the
shape of a torus.
Fig. 22 shows a typical configuration in the smooth phase ($\lambda = 2.0$).
Fig. 23 shows a configuration of the system near the crumpling
transition ($\lambda = 1.5$).
As $\lambda$ is decreased, the the surface becomes more crumpled, as in
Fig. 24, at $\lambda = 1.3$.
Finally in Fig. 25 we show a typical configuration in the crumpled phase
($\lambda = 0.8$).


\bigskip
\noindent
(2) Animation
\medskip

We also developed a module which enables us to visualize the
system by animating the simulation, rather than just displaying individual
meshes. This is done by running the random surface program on the IBM RS/6000
workstation, and using a Unix pipe to pass the data to an AVS module which
we have written to take the data and convert it into AVS format for rendering
and real-time display. We have also included a pause option to freeze the
image at any time. In order to display the results of the simulation in
animated form, we need the power of the IBM RS/6000 to both run the simulation
fast enough, and to do the rendering of the complex surfaces extremely
rapidly. The IBM RS/6000 is not only much faster than the DEC or Sun
workstations which also support AVS, but has a much larger range of colors,
and can use hardware rendering to make the
animation much faster and smoother.

We made a video of this animation, which is invaluable in helping to
explain random surface simulations when making presentations of our
work at conferences and seminars. This animation capability allows us
to better study the dynamics of the system, in particular the change
in the surface when passing through the crumpling transition. It is
also a great help in trying to understand the causes of critical
slowing down, and should help us in our attempts to find new and
better Monte Carlo algorithms.

\bigskip
\noindent
(3) Using AVS to Display Parallel Simulations on a Network
\medskip

The most recent version of AVS provides a parallel module execution function.
We are therefore able to visualize different simulations running on different
machines in the network in parallel.
Here the amount of data being received, and the amount of rendering required
for multiple images, is so large that we need a very powerful machine
such as the IBM RS/6000 for this to be useful. An example of the RS/6000
monitor when running AVS in this fashion is shown in Fig. 26.


%\include{secapp}
%\include{secref}
%  $Id: secref.tex,v 1.8 1994/06/25 13:44:06 han Exp $

\newpage



\newpage

\section*{Figure Captions}
\addcontentsline{toc}{section}{Figure Captions and Figures}

\begin{figure} 
\vspace{8cm}
\protect\caption{
one dimensional string traces out a two dimensional surface in space-time.
}
\label{FIG1}
\end{figure}

=======paper 1=============

  \begin{itemize}

    \item[Fig. 1] The edge curvature $S_E$ as a function of $\lambda$.
As in all other pictures, filled circles and a dotted line correspond to
$N=144$, crosses and
a dashed line indicate $N=288$, and empty squares and a solid line
represent $N=576$.

    \item[Fig. 2] The edge curvature specific heat, $C(\lambda)$.

    \item[Fig. 3] As in Fig. 2, but with the multi-histogram
    reconstruction in
the transient region.

    \item[Fig. 4a] The gyration radius $R_G$ defined in
(\ref{E_RG}), plotted as in Fig. 1.

    \item[Fig. 4b] The effective inverse Hausdorff dimension $\nu$ as a
function of $\lambda$, as defined in (\ref{E_RGD}). The filled dots and the
dashed curve are from a fit to the $N=288$ and $N=144$ data, while the empty
dots and solid curve represent the fit to $N=576$ and $N=288$.

    \item[Fig. 5] The extrinsic Gaussian curvature
$\mid{\cal{K}}\mid$ defined in (\ref{E_CK}), plotted as in Fig. 1.

    \item[Fig. 6] The fluctuations of $\mid {\cal{K}} \mid$.

    \item[Fig. 7] The  intrinsic curvature
$\mid{\cal{R}}\mid$ defined in (\ref{E_CR}), plotted as in Fig.1.

    \item[Fig. 8] The fluctuations of $\mid {\cal{R}} \mid$.

    \item[Fig. 9] The intrinsic extrinsic curvature
correlation, as defined in (\ref{E_KR}), plotted as in Fig.1.

    \item[Fig. 10a] The average maximum coordination
number of the surface vertices,  ${\rm{max}}_i \ q_i$, plotted as in
Fig.1.

    \item[Fig. 10b] The scaling exponent of
${\rm{max}}_i\ q_i$, plotted as in Fig.1.

    \item[Fig. 11a] $S_E$ as a function of Monte Carlo time ($80,000$
steps) for $N=144$, $\lambda=1.4$.

    \item[Fig. 11b] $R$ as a function of Monte Carlo time ($300,000$
steps) for $N=144$, $\lambda=1.4$.

    \item[Fig. 12a] $S_E$ as a function of Monte Carlo time ($80,000$
steps) for $N=144$, $\lambda=1.5$.

    \item[Fig. 12b] $R$ as a function of Monte Carlo time ($300,000$
steps) for $N=144$, $\lambda=1.5$.

    \item[Fig. 13a] $S_E$ as a function of Monte Carlo time ($300,000$
steps) for $N=576$, $\lambda=1.4$.

    \item[Fig. 13b] $R$ as a function of Monte Carlo time ($3,000,000$
steps) for $N=576$, $\lambda=1.4$.

    \item[Fig. 14] $S_E$ and $R$ from the same Monte Carlo run, $N=576$,
$\lambda=1.325$, $20,000$ steps.

\end{itemize}

=======paper 2=============

  \begin{itemize}

    \item[Fig. 1] The edge extrinsic-curvature specific heat
$C(\lambda)$ as a function of
$\lambda$. Multi-histogram reconstructions with errors are shown for $N=576$
(long and short-dashed lines) and $N=1,152$ (solid lines).
Four individual data points are also shown for $N=2,304$ (solid circles).
One sees that the specific heat peak has saturated - it is not growing
with the system size $N$ above $576$.

    \item[Fig. 2] The specific heat $C(\beta)$  of
the two-dimensional $O(3)$ non-linear sigma model as a function of
$\beta$ for lattice volumes $N=16$, $25$, $64$, $100$, $900$, $2,500$,
$4,900$ and $10,000$. The peak saturates
quickly for $N\ge 100$ and ``$\beta_c$'' does not increase with the volume.
  \end{itemize}

\end{document}
